

City Research Online

City, University of London Institutional Repository

Citation: Roszkowska, P., Langer, L. K. & Langer, P. B. (2021). Pension funds and IPO pricing. Evidence from a quasi-experiment. The British Accounting Review, 53(4), 100943. doi: 10.1016/j.bar.2020.100943

This is the accepted version of the paper.

This version of the publication may differ from the final published version.

Permanent repository link: https://openaccess.city.ac.uk/id/eprint/32569/

Link to published version: https://doi.org/10.1016/j.bar.2020.100943

Copyright: City Research Online aims to make research outputs of City, University of London available to a wider audience. Copyright and Moral Rights remain with the author(s) and/or copyright holders. URLs from City Research Online may be freely distributed and linked to.

Reuse: Copies of full items can be used for personal research or study, educational, or not-for-profit purposes without prior permission or charge. Provided that the authors, title and full bibliographic details are credited, a hyperlink and/or URL is given for the original metadata page and the content is not changed in any way.
 City Research Online:
 http://openaccess.city.ac.uk/
 publications@city.ac.uk

Pension Funds and IPO Pricing. Evidence from a Quasi-Experiment

Paulina Roszkowska^{a, c, e,*}, Lukasz K. Langer^b & Piotr B. Langer^d

^a Cass Business School, City, University of London, United Kingdom
 ^b Haas School of Business, University of California, Berkeley, United States
 ^c Hult International Business School, United Kingdom
 ^d Imperial College Business School, Imperial College London, United Kingdom (student)
 ^e SGH Warsaw School of Economics, Poland

Abstract

We exploit a quasi-experiment arising from the government-forced changes to the assets under management and investment policy of the Polish pension funds. We test whether this new regulation and its resultant demand shock on the investors' side, leads to changes in the IPO pricing and the subsequent stock's performance. We report material and a statistically significant decrease in the IPO proceeds (IPO size) in the post-treatment period equal to over 107 million PLN (34 million USD). We find no empirical evidence that the treatment had a significant effect on the first-day IPO underpricing or on the long-term underperformance. We conclude that the demand shock resulting from the pension system reform that primarily aimed at solving fiscal problems effectively eliminated the so-called 'pension premium' of higher IPO valuations. Thus, it indirectly impaired companies' power of raising money in the public stock market. Furthermore, we report a decrease in the average first-day IPO returns among big issuers that is consistent with the book building literature.

Keywords: IPO pricing, first-day returns, long-term underperformance, quasi-experiment, event study, pension funds

JEL codes: G11, G14

^{*} Corresponding author: Paulina Roszkowska, proszkowska@gmail.com, <u>paulina.roszkowska@city.ac.uk</u>, 106 Bunhill Row, London EC1Y 8TZ, United Kingdom.

We greatly appreciate helpful discussions and comments of Panayiotis Andreou, Maximilian Auffhammer, Tomasz Berent, David Blake, Sonia Falconieri, Troup Howard, Ian Marsh, Tomasz Mrowczyk, Terry Odean, Georgios Panos, Panos Patatoukas, Jay Ritter, Wojciech Szwankowski, Young Yoon, and Marcin Żółtek. We thank the Editor and three anonymous Reviewers for their valuable feedback. We also thank the seminar participants at the Haas School of Business and the Warsaw School of Economics, along with conference participants of the International Symposium in Finance-ISF 2018 (Greece), 26th Finance Forum (Spain), ERFIN 2018 (Warsaw), and Financial Management & Accounting Research Conference 2019 (Cyprus) for their helpful comments.

1. Introduction

Effective February 2014, the Polish government introduced new legislation that regulated the allocation of assets under management (hereafter, 'AUM') and the investment policy of the Polish pension funds. The most significant changes encompassed a lower contribution to private pension funds, the gradual transfer of funds related to persons approaching the retirement age, and a one-off transfer of all government-guaranteed securities from private pension funds to the state-governed administrator. The main reason for these changes was to aid the unsustainable pay-as-you-go state-governed part of the pension system in Poland. Moreover, the new regulation restrained pension funds from high-risk equity investments, but at the same time loosened their investment policy by eliminating the internal benchmarking mechanism as a measure of fund's performance. In consequence, we observe lower net cash flows to pension funds after the 2014 reform.

A long line of literature documents the effects of pension funds' investing process on capital markets. One stream of research finds that the investment activity of pension funds is beneficial to equity markets by promoting liquidity, boosting demand for capital-market instruments and innovation, and contributing to the development of capital markets on the whole (Davis, 1998; Walker & Lefort, 2000). Conversely, Singh (1996) and Roldos (2004) suggest that institutional investors might exert a negative influence on the capital markets in emerging economies. In particular, private pension funds' investments contribute to asset price distortions, bubbles, and concentration of risks (Roldos, 2004). In the context of the Polish Pension Reform of 1999, Zalewska (2006) shows that the benefits of pension funds' investment in the home market are only short-term and revert in the long run. Despite the intense debate on the capital market effects of pension fund investments and pension system reforms, none of the papers specifically looks at how they affect the new listings and their aftermarket performance in equity markets.

In this paper, we investigate whether there is a link between the pension fund reform of 2014 and the pricing of initial public offerings (hereafter, 'IPO'). We exploit a quasi-experiment arising from the exogenous treatment of the Polish pension funds to test whether a demand shock on the investors' side leads to changes in the IPO market. We argue that lower net cash flows to pension funds and eliminating the internal benchmark mechanism as a measure of fund's performance (the leading cause of herding among the Polish pension funds in the pre-reform time) decreased funds' aggregate demand for equities, and specifically for new listings. Our research idea is motivated by the practitioners' claim that this regulatory change could have had "a negative impact on the Warsaw Stock Exchange, where the funds have been crucial

buyers of shares and have been responsible for the so-called 'pension premium' of higher valuations" (Cienski, 2014). Specifically, we intend to test whether a demand shock on the investors' side, caused by the regulatory change, led to changes in the size of the IPO proceeds and the level of the IPO underpricing.

We study IPOs in the Warsaw Stock Exchange Main Market (hereafter, 'WSE'), which has operated since 1991. Importantly, we also look at the NewConnect¹ (hereafter, 'NC')—an alternative stock market created in 2007 to aid smaller entities that could not meet the requirements (either capital, procedural, or disclosure) of the WSE. As of June 30, 2017, both platforms included 891 entities with a combined market capitalisation of 1,326 billion PLN, which was equivalent to 358 billion USD² (GPW, 2017; NewConnect, 2017). Noteworthy, years of reforms, restructuring, and privatisation of the formerly state-owned companies made it possible to create the largest and the most important stock bourse in Central and Eastern Europe, characterised by the largest IPOs, improved liquidity, transparency and investor protection (Jackowicz, Kowalewski, Kozłowski, & Roszkowska, 2017). In 2016 and 2017, the Warsaw Stock Exchange reported the third-highest number of IPOs among all European exchanges (PwC, 2018). Further, the Polish stock exchange was recently classified by the FTSE Russell as a 'developed market' in its FTSE Global Equity Index Series—an improvement from the 'advanced emerging' category (FTSE, 2017). Despite its regional importance, research on the IPO effects in the Polish stock market is still scarce.

We use a unique dataset of IPOs listed in the two equity markets for the years 2005-2017. We combine multiple databases and hand-collect missing data to arrive at the most comprehensive sample ever used for the Polish stock market³. The abundance of observations in our dataset yields an opportunity to investigate IPO underpricing effects in the Polish stock market. We find strong support for both the IPO underpricing and long-term underperformance effects observed previously in the U.S. and other advanced markets.

¹ The NewConnect market is less formal than the main market. It takes less time for a firm to go public there due to simpler and faster conditions for introducing the company to trading, there are lighter information obligations (e.g. the lack of audited semi-annual reports) and low debut costs. However, the market microstructure is the same in the main and alternative market (e.g. regarding how trade is done, the role of market makers, etc.)

² Exchange rates are as per the official Central Bank of Poland statistics; retrieved on February 23, 2019, from <u>https://www.nbp.pl/homen.aspx?navid=archen&c=/ascx/TabArchAen.ascx&n=17a125en</u> for June 30, 2017.

³ Jelic and Briston (2003) study the average market-adjusted first-day returns and report the average underpricing of 27.37% from 1991 to 1998, while Jewartowski and Lizińska (2012) report the average underpricing of 13.95% for the subsequent ten years of 1998–2008. However, both studies are rather confined because they encompass short periods, a limited number of stocks and they focus only on the main equity market of the Warsaw Stock Exchange.

We evaluate the impact of the government-forced changes to the AUM and investment policy of the Polish pension funds on the IPO pricing and aftermarket performance using various approaches and robustness checks. First, we compare the mean outcomes in the pre- and post-treatment periods, separately for the WSE and NC markets. The two-sample *t*-test provides first insights. We find that the mean IPO size decreased by 142 million PLN (over 45 million USD⁴) in the WSE market post-treatment. At the same time, there was no statistically significant effect in the control (NC) sample. Then, we find no statistically significant changes in either the mean IPO first-day returns or in the mean long-term underperformance. Robustness tests based on bootstrapping confirm our results.

Further, we look for the causal effect between the regulatory treatment of the Polish pension funds and changes in the IPO pricing using the difference-in-differences research design. Such an identification is possible because our data consists of two subsamples: WSE and NC IPOs. Pension funds in Poland only invest in the former market, and not in the latter due to regulatory constraints. Therefore, we can compare the treated (WSE) and nontreated (NC) IPO outcomes, and ascribe any differences to the effect of the regulation. We first report a notable and statistically significant change in the IPO size: average IPO proceeds decrease by almost 107 million PLN after the regulatory treatment of the Polish pension funds. We attribute this finding to the institutional investors' decreased aggregate demand and more (practical) independence in making investment decisions (release from the internal benchmark as a performance measure). We argue that there is a new equilibrium established in the IPO market after the reform where listing companies can raise relatively less money on average (no more 'pension premium'). We discard a possibility of big companies cancelling IPOs for the reason of not being able to raise as much money as before the reform because we see neither a decline in the post-treatment number of new listings nor an increase in the number of IPO withdrawals. To check for the robustness of our results, we employ a placebo test where we alter the date of the event. We run difference-in-differences regressions as in the original specification using the simulated 'event time' as each day from the +/- 3-year period around the original event, and we analyse the *p*-values of the difference-in-differences estimator. Our results are robust to placebo testing. Only around the original date of the regulatory change do the *p*-values fall close to the zero-level. We conclude that a demand shock resulting from the pension fund reform of 2014

⁴ Exchange rates are as per the official Central Bank of Poland statistics; retrieved on February 23, 2019, from <u>https://www.nbp.pl/homen.aspx?navid=archen&c=/ascx/TabArchAen.ascx&n=14a022en</u>, for February 3, 2014.

effectively decreased the benefits of newly listed companies regarding the amount of money raised in the public stock market.

Additionally, our cross-sectional tests based on IPO size reveal that the effects of regulation are concentrated among larger issuers. This observation is consistent with a blue-chip bias, whereby Polish pension funds prioritize larger firms in their equity portfolios (Zalewska, 2006). The effect of the reform on the IPO size is statistically significant in the larger two IPO size quantiles, but not in the smaller two quantiles. Moreover, we report a statistically significant decrease in the IPO underpricing. This decrease is because underpricing is a means of compensating institutional investors for the critical role they play in supporting IPOs. First, the book building literature posits that IPO underpricing is used by investment bankers to elicit revelation of information about the market value of an IPO during the book building process (e.g., Benveniste & Spindt, 1989). Second, underwriters use IPO underpricing to further compensate institutional investors for supporting IPOs with weaker post-issue demand. Hereto, institutional investors support IPOs by agreeing to hold their allocations for a longer period (Chemmanur, Hu, & Huang, 2010). Against this backdrop, we expect to see a decline in the level of underpricing in response to the above-described demand shock on the institutional investor side. Indeed, for the largest IPO quartile, where Polish pension funds primarily invest, we see a statistically significant decline in the first-day returns following the reform. We conclude that the reform not only affected the amount of pension funds' AUM, but it also had a detrimental effect on how much they make on average when investing in IPOs as the first-day returns of large IPOs decreased.

The contribution of this paper is three-fold. First, our paper provides empirical evidence as to whether the IPO underpricing and long-term underperformance effects are present in each of the equity markets in Poland. Second, by exploiting a quasi-experiment, we shed new light on the indirect impacts of the 2014 pension funds reform. We show that the decreasing net cash flows to pension funds AUM and abandoning the internal benchmark as a measure of funds' performance (a shock on the demand side) result in lower IPO proceeds. At the same time, we report a significant decrease in the first-day returns for the subsample of the largest issuers, consistent with book building theories of IPO underpricing. These findings are of interest to both market participants but also legislators if they consider similar reforms in the future. Lastly, by providing insights about the changed levels of IPO first-day returns and IPO long-term underperformance in the reform aftermath, we contribute to the literature on the factors that drive IPO underpricing effects and to the literature on asset pricing reaction to demand shocks

on the investor side. We argue that the insights and contributions provided in this study are of importance to the international community, given the scale of IPO activity in the Warsaw Stock Exchange. Our tests have been run on a sample of 706 IPOs for the years 2008-2017, while in the U.S.—the biggest stock market in the world—there have been 990 IPOs in a respective period (Ritter, 2019).

The rest of the paper is organised as follows. Section 2 depicts the regulatory change to the pension fund AUM and investment policy. In Section 3, we develop research hypotheses. Section 4 explains the research methodology. Section 5 details the data and provides descriptive statistics. Section 6 presents the main results and robustness tests. Section 7 concludes.

2. Pension Fund Reform of 2014: The Setting

The contemporary version of the Polish pension system follows the World Bank's multi-pillar design since its establishment in 1999 (Holzmann, 2000). Voronkova and Bohl (2005) provide a comprehensive description of the Polish pension fund sector, which consists of the two obligatory pillars: (1) a pay-as-you-go system managed by a state-owned entity (the Social Insurance Institution), (2) open pension funds run by private managing companies, and one voluntary pillar: (3) privately funded pension security schemes.

In time, however, this pension system demonstrated severe shortcomings. Most importantly, the system turned out to be unsustainable, which manifested itself in the growing fiscal problem of not enough funds allocated to the first pillar for covering state pension obligations. System unsustainability was the main reason for the 2014 policy change that resulted in several alterations to the pension system. The key changes included: (i) a one-off asset transfer of Tbonds (written-off immediately), government-guaranteed bonds, municipal bonds, and cash in the amount of 51.5% of their total assets from the second pillar (pension funds AUM) to the first pillar; (ii) lower contribution rate to the second pillar: 2.92 per cent instead of 3.5 per cent of income, and (iii) gradual transfer of assets of persons ten years before the retirement age from the second pillar to the first pillar (IMF, 2014; Journal of Laws 2013, Item 1717, 2013). The legal act was signed in December 2013, while legal changes were enforced on February 3, 2014. The one-off transfer took effect on the same date. From the pensioners' standpoint, the value of their contributions only changed hands from privately-managed to a state-managed institution. The transfer of debt securities did not have any effect on pension funds' equity investments either. However, future contributions and AUM were to decrease, which could affect equity investments. Moreover, effective February 2014, pension funds were restricted from investing in government-backed securities, while a minimum of 75% of their assets under management was to be invested in equities (with a ban to invest in the NC market).

Additionally, until July 2014 pension funds were required to guarantee a minimum rate of return on their investments, which was set every quarter based on the average performance of all pension funds (an internal benchmark mechanism). In case of not meeting the minimum return, a fund covered the shortfall from its special reserves, own assets, joint pension funds' reserves, or from the government's funds in the last resort. The reason for imposing such investment policy restrictions on pension funds was to protect informed and uninformed contributors (KNUiFE, 2000). This mechanism gave rise to herding. Voronkova and Bohl (2005) and Kominek (2012) document strong herding behaviour by Polish pension funds, much more intense than in other developed markets, and particularly evident regarding investment in largecapitalization stocks. In July 2014, the internal benchmarking mechanism stopped serving as a performance measure for the Polish pension funds. Thus, the herding should no longer exist and should not determine pension funds' investments.

3. Background and Hypotheses Development

Our hypotheses are first motivated by macro-finance literature. Roldos (2004) underlines the importance of loosening the regulatory framework for pension funds. Due to herding practices among these entities, changes in investing sentiment usually result in widespread changes. Therefore, even a marginal policy change may have a substantial effect on the capital market as a whole. In the Polish context, policymakers prioritise political and social outcomes of pension fund reforms (Góra, 2014). The caveat of such behaviour is that it potentially comes at the expense of capital markets prosperity. Up to date, the literature did not consider how these policy changes may affect IPOs and thus the private sector.

3.1. Pension Fund Reform and IPO Pricing

A few theoretical developments and stylized facts motivate us to study the relationship between the described regulatory change and IPO pricing. Primarily, we are intrigued by untested practitioners' claims that after the regulatory change of 2014, pension funds started pushing for a reduction in the price of offers (Kucharczyk, 2016) and that post-reform there should no longer be a 'pension premium' of higher valuations in the WSE (Cienski, 2014).

We argue that there are two specific elements of the regulatory change that drive such expectations. First, as a result of the lower contribution rate to the second pillar and a gradual

transfer of pension assets of persons ten years before the retirement age to the first pillar, there would be a decrease in the net cash inflow to the open pension funds. Coval and Stafford (2007) provide evidence that changes in capital flows to institutional investors' AUM create price pressures in equity markets. Funds that experience significant outflows have no choice but to sell some of their holdings to cover redemptions ('asset fire sales'). When many funds sell at the same time, one should expect a negative price pressure in the market. We anticipate this mechanism to work also in our setting: reduced inflows to all pension funds decrease their aggregate demand for new shares, which negatively affects the amount of money raised by listing companies.⁵

Our expectation is conditional on pension funds having economic power enough to affect equity prices. As reported in the stock exchange statistics, the second pillar—open pension funds— has been a significant equity investor in the Warsaw Stock Exchange. Between end-2013 (around the regulatory change) and mid-2017 (end of our sample), pension funds' equity investments have been growing from 100 billion PLN (about 33 billion USD) to over 148 billion PLN (nearly 40 billion USD)⁶. This corresponds to almost 12% share of the joint WSE and NC market capitalization end-2013, and 11.2% end-2017 (GPW, 2018). More importantly, Polish pension funds are a repeated 'price maker' during the book building process, determining the ultimate offering price. Zalewska (2006) provides evidence that pension funds take significant stakes in the new listings⁷. Hereto, there is an apparent blue-chip bias: the average free float taken over by the pension funds in the highest IPO-size quartile (large cap companies) equals to 36.4%, successively decreasing to 26% in the lowest IPO size quartile (small cap companies), which is still a significant stake being taken up by only a few institutions. Additionally, as confirmed in our consultations with pension fund managers and investment bankers,⁸ pension

⁵ In addition, Chemmanur et al. (2010) argue that pension funds are contributing to both the demand at the time of the issue and the post-issue demand.

⁶ Exchange rates are as per the official Central Bank of Poland statistics; retrieved on February 23, 2019, from <u>https://www.nbp.pl/homen.aspx?navid=archen&c=/ascx/TabArchAen.ascx&n=13a251en</u> for December 31, 2013, and from <u>https://www.nbp.pl/homen.aspx?navid=archen&c=/ascx/TabArchAen.ascx&n=17a125en</u> for June 30, 2017.

⁷ Zalewska (2006) also gives examples of IPOs, where Polish pension funds jointly took over 60% of the newly issued shares, e.g., 12 open pension funds absorbed 63.8% of the free float of GTC's IPO in 2004.

⁸ In the research process, we have consulted our hypotheses and results with pension fund managers in Poland. Consultations have been made in form of phone calls conducted between October 2018 and January 2019, and again between November and December 2019. The interviewees were asked unstructured questions about the practice of their investing, about the effect of the reform on their trading decisions, and about their perception of the reform and its outcome on the equity market in Poland. There has been no specific questionnaire followed in those interviews. The consulted fund managers represent four of the seven biggest pension funds in Poland (AUM-wise), and all wished to remain anonymous. We refer to practitioners' insights a few more times throughout the paper.

funds are the first subject to be approached during the roadshow when an investment bank assesses the overall demand for the shares of the new issue.

Second, the departure from the minimum rate of return as a measure of performance reduces incentives for herding. Blake and Timmermann (2002) suggest that when the evaluation benchmark for fund managers is a weighted average (as it used to be in Poland before 2014), the safest investment strategy is to follow market leaders. Lakonishok, Shleifer, and Vishny (1992) document herding behaviour in a large sample of U.S. pension funds. Similar findings have been reported for the behaviour of the Polish pension funds before 2014 (Kominek, 2012; Voronkova & Bohl, 2005). Zalewska (2006) provides indirect evidence for herding while taking up shares of new listees, especially in cases of big IPOs. She reports that out of 15 open pension funds active in Poland, the average number of pension funds investing in the new shares of the companies from the highest IPO size quartile was 11.4, and it decreased successively towards 2.6 in the lowest quartile. This decline is in line with fund managers' accounts of their behaviour in the primary market: before the regulatory change, they tried to mimic each other's equity engagements in new, especially more prominent listees to stay close to the internal benchmark with their portfolio performance. In practitioner's views, by declaring demand for shares of a listing company primarily because other funds declared their demand, a fund contributes to an artificial 'pension premium.'

Marrying these two immediate outcomes of the 2014 pension fund reform: the decrease in net cash flows to pension funds with eliminating the main driver for herding behaviour, we argue that the regulation, aimed initially at covering the state deficit in the pension system, was also a negative demand shock for equities in the primary stock market. Our premise is that as a result of this negative shock on the investors' side, pension funds bid less during the book building process⁹ (no more 'pension premium') and, as a result, companies raise less money at an IPO. We test our prediction with the following hypothesis:

Hypothesis 1:Government-forced changes to the AUM and investment policy
of the Polish pension decreased the average IPO size.

⁹ Noteworthy, the IPO process in the Warsaw Stock Exchange follows typical characteristics and stages of IPOs in other markets, for instance in the U.S. as explained by Ellis et al. (2000) or Brau and Fawcett (2006). The main practical difference is that in the U.S., most IPOs are done within a firm-commitment agreement with the underwriter (the underwriter purchases the whole offer and resells the shares to investors, while the issuing company is guaranteed that a particular sum of money will be raised), while in Poland, most transactions happen on the best efforts agreement (the underwriter does not guarantee the amount raised for the issuing company, rather it's role is limited to sells the securities on behalf of the issuing company).

We state the alternative hypothesis that government-forced changes to the AUM and investment policy of the Polish pension funds did not affect the average IPO size.¹⁰ Failing to reject the null hypothesis will imply that lower net cash flows to pension funds and eliminating formal incentives for herding behaviour reduced the amount of money listing companies raise on average at an IPO. The counterfactual reasoning, in this case, is that the event was not meaningful for the money raised at an IPO. Thus, IPO size did not change after the event. Accordingly, the time trend of the average yearly IPO size in the WSE post-reform should be mirroring the time trend of an average annual IPO size in the NC market. We do not envision a positive scenario, under which the average IPO size increases in the reform aftermath, as this would be both surprising and counterintuitive. There is no empirical evidence suggesting that after a demand shock on the investor side, equities were priced higher on any stock exchange.

3.2. Pension Fund Reform and IPO Underpricing Effects

In a seminal study, Stoll and Curley (1970) observe a 'remarkable price appreciation' between the IPO offering price and the first-day closing market price. Soon after that, Logue (1973) and Ibbotson and Jaffe (1975) provide empirical evidence for this phenomenon that is currently known as the 'IPO underpricing effect'—a situation when a stock is priced below the value anticipated by the market, leading to abnormal returns in the first day of trading. Recent evidence suggests that at an aggregated level, stocks listed in the U.S. exchanges experienced an average first-day price appreciation of 7.2%, 14.8%, 64.6%, and 14% during the periods of 1980–1989, 1990–1998, 1999–2000 and 2001–2016, respectively (Ritter, 2017). Similar IPO underpricing is observed in other capital markets.

In this paper, our investigation extends onto examining whether the exogenous treatment of pension funds that originated a demand shock on the investors' side, caused any changes in the level of the first-day IPO returns in the stock market of Poland. Our analysis is motivated by the fact that pension funds are materially invested in IPOs, especially in the most significant IPOs. If the 2014 pension reform was also a negative demand shock for equities in the secondary stock market, then the lower activity of institutional investors could lead to higher first-day returns due to sentiment (Ljungqvist, Nanda, & Singh, 2006) and information asymmetries

¹⁰ Because the two changes coincided in time and IPO observations are only observed at discrete intervals, it is impossible to test for the two driving forces separately.

(Field & Lowry, 2009)¹¹. Alternatively, any decline in the first-day returns would be consistent with the book building literature (Aggarwal, Prabhala, & Puri, 2002; Benveniste & Spindt, 1989; Chemmanur et al., 2010; Rock, 1986) that posits that IPO underpricing is a mean of compensating institutional investors for truthfully disclosing value-relevant information during the book building process. Specifically, institutional investors have access to private information on the fair IPO value, and investment bankers use underpricing as a tool of eliciting that information during the pre-selling period. Accordingly, there is a positive association between the level of underpricing and the level of institutional investors' participation in the book building process. In our setting, this story can be recast through the following mechanism: in the pre-reform equilibrium, ceteris paribus, there was a certain average level of IPO underpricing that was 'used' by underwriters to compensate institutional investors. In the immediate post-reform period, however, there would be less underpricing as a larger fraction of the market is taken up by retail investors, who are less likely to possess and disclose valuerelevant information, and who do not need to be compensated with extra underpricing. Given these two potential alternative scenarios, the ultimate effect of the negative demand shock on the investors' side on the IPO underpricing is *ex-ante* unclear. Therefore, we state the following null hypothesis pertaining to the aftermarket first-day IPO returns as:

Hypothesis 2: Government-forced changes to the AUM and investment policy of the Polish pension funds did not affect the degree of the firstday IPO underpricing.

The null hypothesis is consistent with the reasoning that pension funds are not affecting the first-day closing price in the treatment period to counterbalance the consequences of the reform because, in general, they invest for a long-term horizon and, according to practitioners, they do not rebalance their portfolios with newly listed shares shortly after an IPO (especially during the first day of trade). If the null hypothesis is rejected, the sign of the estimated relationship could be either positive or negative due to offsetting mechanisms affecting the first-day returns. Accordingly, the alternative hypothesis states that the government-forced changes to the AUM and investment policy of the Polish pension funds affected the first-day IPO returns.

After the first-day abnormal price appreciation, there is a tendency for the subsequent downward-oriented stock price adjustment (Miller, 1977). Ritter (1991) names this

¹¹ Ljungqvist et al. (2006) argue that the smaller the involvement of institutional investors, the more sentimentdriven investment decisions, and thus higher underpricing. Field and Lowry (2009) add that institutional investors have an information advantage, therefore due to their investments, stocks tend to underperform less in the longterm.

phenomenon an 'IPO underperformance effect' and defines it as the stock's underperformance in the years following an IPO. Miller (1977) explains this tendency through the combination of heterogeneous expectations (pessimistic vs optimistic investors) and short-sales constraints. In this paper, we investigate whether the regulatory change of 2014 affected the level of the IPO underperformance. The decrease in the activity of institutional investors after the treatment could exacerbate the IPO underperformance by leaving more room for sentiment driven investment and information asymmetries (Field & Lowry, 2009). Thereupon, we state the following null hypothesis pertaining to the effect of the reform on the IPO underperformance within a short-term (30-day) time window following an IPO:

Hypothesis 3:Government-forced changes to the AUM and investment policy
of the Polish pension funds did not affect the degree of short-
term IPO underperformance.

The alternative hypothesis states that the government-forced changes to the AUM and investment policy of the Polish pension funds did affect the short-term underperformance. We do not specify the counterfactual more precisely because the short-term window also includes the first-day returns.

Notably, the positive association between institutional trading of IPOs in the aftermarket and the long-term IPO performance decays over time (Chemmanur et al., 2010). This is because after the company goes public, a substantial amount of new information is provided to the market, which reduces the information advantage of institutional investors. Accordingly, we state the following null hypothesis testing for the effect of the reform on the IPO underperformance within a short-term (30-day) time window following an IPO:

Hypothesis 4:Government-forced changes to the AUM and investment policy
of the Polish pension funds did not affect the degree of long-term
IPO underperformance.

The counterfactual reasoning could be that the 2014 reform did affect the degree of the longterm IPO underpricing. Still, we do not find previous empirical evidence in support of the alternative hypothesis.

Lastly, we explore a theoretical possibility of an alternative explanation of our results: given the new market conditions, where pension funds have less money to invest, firms' managers evaluate the potential IPO proceeds against the costs of raising capital, and because they anticipate lower demand for their company's shares (fear of not raising enough funds), they choose not to do an IPO at all (Lowry, 2003). Firms that still decide to go public under new conditions would need to 'compensate' investors for the unfavourable financial environment with higher underpricing. For example, Dorsman and Gaunopoulos (2013) show that the European sovereign crisis resulted in Dutch IPOs 'leaving more money on the table'.

Our identification relies on the specific design of the Polish stock market. As previously described, the government-forced changes to pension funds' AUM and investment policy serve as an exogenous shock to IPO pricing and stocks' subsequent performance. Importantly, this is only true for one of our markets under research: the WSE, because pension funds do not invest in the NC market. Before 2014, Polish pension funds' investments in the NC companies were close to zero, while after 2014, their investment in the alternative market was forbidden. Thus, the shock originating from the regulatory change had the potential to affect IPO pricing and stock's subsequent performance only in the WSE subsample. By exploiting a quasi-experiment, we can test for the changes in IPO pricing and the degree of IPO underpricing effects as a consequence of the 2014 regulatory change.

4. Methodology

To test for IPO underpricing and long-term underperformance, we calculate the first-day IPO returns and post-IPO 30-day and 360-day returns. For *n*-th stock, we measure the first-day return ($RET_{n,1}$) as the difference between the closing price of the first day of trading, $P_{n,1}$, and the IPO offer price, $P_{n,0}$, divided by that closing price, as in Equation (1):

$$RET_{n,1} = \frac{P_{n,1} - P_{n,0}}{P_{n,0}}.$$
(1)

We calculate short- and long-term returns as compounded stock price return over a *k*-day period following the equity issuance (i.e. for 30 and 360 calendar days, respectively). As in Barber and Lyon (1997), to calculate the non-market-adjusted returns for the period starting at the IPO date (t=1) and ending after *k* days, we use the buy-and-hold (holding period) rate of return (*HPR*_{*n,k*}), as in Equation (2):

$$HPR_{n,k} = \prod_{t=1}^{k} (1 + RET_{n,t}) - 1,$$
(2)

where $RET_{n,t}$ is the return on day *t* for stock *n*, calculated using adjusted prices. In the average long-term cumulated return calculations, an individual stock forms a single portfolio, and both

size- and equal-weights are used. If any data is missing for a company, we exclude it from calculations for all research periods starting with the first missing data point.

To evaluate performance, we adjust returns using the Capital Asset Pricing Model (CAPM) market factor. The CAPM-assumed market portfolio does not occur in reality, and its betas, as systematic risk measures, are not sufficient to explain the cross-section of expected returns (Roll, 1977). Additionally, the model's empirical robustness is questionable (see Roszkowska and Langer (2016b) for evidence of its poor performance in the Polish stock market and Fama and French (2015) for the similar conclusions for the global markets). Nonetheless, a single-factor model allows for convenient estimation of risk-adjusted returns in an everyday setting. Thus, we calculate the abnormal return for an *n*-th stock for each day k ($AR_{n,k}$) as in Equation (3):

$$AR_{n,k} = RET_{n,k} - [Rf + \beta_j \times MKT_n], \qquad (3)$$

where Rf is the risk-free rate, β_j is the respective industry *j*-th beta, and MKT_n is the single market risk factor premium for the *n*-th stock. Then, following Ritter (1991), we compute the cumulative abnormal returns (*CAR*). For the *n*-th stock and the period from the IPO date (*t*) to the *k*th day after an IPO, we calculate abnormal return as in Equation (4):

$$CAR_{n,k} = \sum_{t=1}^{k} AR_t.$$
(4)

The one-month Warsaw Interbank Offer Rate (WIBOR) proxies the risk-free rate, given the lack of data on short-term treasury bills in Poland. Roszkowska and Langer (2016a) test for the merits of using WIBOR as the risk-free rate and find no significant differences between WIBOR and the periodically available time series of short-term treasury bills. For the beta coefficients, we use average levered industry beta estimates for the period of 2011–2016 from the Damodaran database (Damodaran, 2018). We use this database for Europe as it includes Polish stocks. We proxy the market risk factor premium using the adjusted return on the Warsaw Stock Exchange Index (WIG). This total return index includes dividends and pre-emptive rights of all companies listed in the primary market, excluding foreign companies and investment funds.

Finally, we study the impact of an exogenous shock in the form of government-forced changes to pension funds' AUM and investment policy on the IPO pricing. We begin by testing for the differences in mean IPO returns (first-day returns, short- and long-term performance) and mean IPO size between the pre- and post-treatment subsamples. Hereto, we use a two-sample *t*-test.

We split the sample by the IPO date, using February 3, 2014, as a cut-off date. We separately test for the differences in means in the WSE and NC subsamples. We do not log-transform IPO size because we expect the results to be driven by large IPOs, where pension funds heavily invest as confirmed by our interviewing of pension fund managers, and as previously evidenced by Zalewska (2006).

We test for the robustness of these results using a nonparametric bootstrap methodology. In doing so, we address (and try to eliminate) the possibility that our results are due to chance, i.e. smaller IPOs randomly occur in the post-treatment period, in the treated sample. To generate possible outcomes due to chance, we randomly sample IPOs with replacement and calculate the difference between mean IPO size in the pre-treatment period and mean IPO size in the post-treatment period To test for significance of the differences we use the *p*-value calculated based on the bootstrap sample. Specifically, we calculate *p*-value as in Equation (5):

$$p_{boot} = \frac{1}{N} \sum_{i=1}^{N} I(d_i > \bar{d})$$
⁽⁵⁾

where N is the number of bootstrap iterations, d_i is the pre-post difference in the iteration *i*, and \overline{d} is the observed pre-post difference.

We use February 3, 2014, as our cut-off date, and the symmetric 3-year period around the cutoff when sampling the IPOs (short-time window). The number of IPOs in every bootstrap sample, in both pre- and post-intervention periods, matches the total number of IPOs in our sample, in both pre- and post-intervention periods. We report the density based on 10,000 iterations, and compare the differences estimated in our sample to the distribution of differences based on the sampling results. We repeat the same procedure separately for the treated (WSE) and control (NC) group.

Next, we implement the difference-in-differences regression methodology. This methodology compares the effect of an event on groups affected thereby (treated) with those that are unaffected (control). The treated group corresponds to the WSE new listings, in which pension funds (directly affected by the legal change) are materially invested, and the control group corresponds to the NC new listings, in which pension funds do not invest at all. Thus, any change to the pension funds' AUM or investment policy could only have a potential effect on the IPO pricing in the WSE, and not in the NC market. Noteworthy, the difference-in-differences strategy eliminates the bias that comes from other unobserved changes that could have affected the treated group.

Accordingly, using IPO-level data, we study the response to the exogenous shock regulatory change using the cross-sectional regression specification as in Equation (6):

$$Y_i = \beta_1 WSE_i + \beta_2 WSE_i \times Post_i + \gamma_t + \delta_i + \varepsilon_i.$$
(6)

The dependent variable Y_i is either the first-day return, short- or long-term underperformance, or the IPO Size for the *i*-th stock. WSE_i is a binary indicator equal to 1 for the IPO listings in the treated WSE market, and 0 otherwise. $Post_i$ is a binary indicator equal to 1 for the IPO listings after the pension fund reform came into force, and 0 otherwise. Thus, the interaction term $WSE_i \times Post_i$ captures the difference-in-differences effect. γ_t are the time fixed effects to control for unobservable differences between the IPOs of firms listed in different years, t, and δ_j are the industry fixed effects to control for time-invariant differences between the IPOs of firms from different industries, j. ¹² ε_i is the error term. We omit the $Post_i$ variable because the pre-post firm-invariant differences are subsumed by year fixed effects. ¹³ We also omit firmspecific characteristics because, unlike fixed effects, they can be endogenous to our treatment assignment, and hence bias our coefficient of interest (e.g., Roberts & Whited, 2013). Following Bertrand et al. (2004), we cluster standard errors to avoid overstating the significance of our difference-in-differences estimates. We cluster our errors at industry-year levels to ensure a sufficient number of clusters.

To ensure the robustness of our results, we augment our baseline regression specification with additional control variables. First, we include the natural logarithm of total assets to control for the firm size (e.g., Barth, Landsman, & Taylor, 2017; Liu & Ritter, 2010; Loughran & Ritter, 2004). Second, we include the return on assets (ROA) to control for the firm profitability, or more broadly for the firm quality (e.g., Barth et al., 2017; Chang, Chiang, Qian, & Ritter, 2016). Finally, we also include VC backing status as an additional control (e.g., Barth et al., 2017; Liu & Ritter, 2010; Loughran & Ritter, 2004). Accordingly, we use the regression specification as per Equation (7):

¹² In an additional robustness check, we further add interactions between industry fixed effects and the post indicator to directly control for industry-specific differences between pre- and post-treatment periods that could affect our results. Our main coefficients remain statistically significant. Table C10 in Appendix C reports our results using this alternative specification.

¹³ In the alternative specification of the difference-in-differences model we also consider a simpler model that excludes year fixed effects but includes a post-treatment indicator. Table C6 and Table C11 in Appendix C report our results using this alternative model without and with additional controls, respectively.

$$Y_{i} = \beta_{1}WSE_{i} + \beta_{2}WSE_{i} \times Post_{i} + \beta_{3}log (Assets)_{i} + \beta_{4}ROA_{i} + \beta_{5}I(VC)_{i} + \gamma_{t} + \delta_{j} + \varepsilon_{i}.$$
(7)

log (*Assets*) is the total assets in the last fiscal year before the IPO. This variable controls for the confounding effects attributable to the firm size. *ROA* is the return on assets in the last fiscal year before the IPO, using the net income in the numerator and the end-of-period value of total assets in the denominator. I(VC) is an indicator of whether an IPO was Venture Capital-backed.

We perform our tests using the symmetric three-year window around the cut-off date of February 4, 2014. Thus, we have the pre-treatment period of February 4, 2011, to February 3, 2014, and the post-treatment period of February 4, 2014, to February 3, 2017. We complement our analysis with an implementation strategy that uses long-time window based on the full data, where the pre-treatment period is January 1, 2008, to February 3, 2014 (six years one month), and the post-treatment period is February 4, 2014, to June 30, 2017 (three years five months). Whereas the narrower time window might help establish the relationship on the treatment group, longer series alleviates concerns related to short-term autocorrelation (see Hausman & Rapson, 2018). We trim the time-series in January 2008, due to data availability: the NC market launched in 2007 and pension funds' detailed holdings are available from 2008.¹⁴

To reinforce our result, we run a robustness check in the form of a placebo test. Specifically, we re-estimate our regression based on the different cut-off dates from our sample. We again use a symmetric window of three years, conditional on data availability. In the final analysis, we reassess the significance of the coefficient on the interaction terms depending on the cut-off date.

5. Data and Descriptive Statistics

5.1. Data

We obtain data on IPOs in the Polish stock market from the following providers: *GPW* (gpw.pl), *Stooq* (stooq.pl), *Bossa* (bossa.pl), and *Bloomberg*. *GPW* data include a complete list of IPOs and a summary of the most relevant IPO data (i.e. IPO date, size, and offer price). We retrieve the adjusted daily stock prices from the *Stooq* database. Adjustments account for stock splits, dividends and other distributions, rights offerings, and denominations. We retrieve non-adjusted

¹⁴ Because 2017 is incomplete in our dataset, we also test for the robustness of our results in the sample that excludes the year 2017. We confirm that our results are robust to exclusion of the year 2017. In fact, when we use a long-period estimation window, the coefficient on the $WSE \times Post$ variable is higher in magnitude and statistical significance.

prices from the *Bossa* database. *Bloomberg* is used to obtaining any missing time series data, particularly the daily adjusted and non-adjusted time series of the stock prices for delisted, acquired, and liquidated companies. The missing IPO data is hand-collected from companies' prospectuses. We obtain financial statements data and company identifiers from Notoria. Finally, we obtain additional IPO information from Dealogic. Appendix A provides a detailed description of our data.

Our data spans from January 2008 to June 2017. We trim the time-series, excluding observations prior to January 1, 2008, due to data availability: the NC market was only launched in 2007 and pension funds' detailed holdings are available from 2008. Additionally, Lyn and Zychowicz (2003) provide evidence that the small number of companies, ambiguous macroeconomic conditions, lack of proper structure and legislation, and a shortage of skilled brokers led to numerous pricing abnormalities in the WSE during the 1990s and early 2000s. This could bias the patterns of IPO pricing for the time series before 2008. Another reason for limiting the sample is the anomaly reported by Lyn and Zychowicz (2003): suspiciously high IPO underpricing (54.45%) for a sample of 103 IPOs in Poland from 1991 to 1998 with respectively high standard deviation (82.23%).

Our sample includes 706 IPOs: 237 listings in the WSE and 469 listings in the NC.¹⁵ It is important to note that the sample diminishes in size when we analyse post-IPO short- and long-term underperformance. Some companies lacked the time series of returns after an IPO, either because they were no longer active (e.g., delisting within the first year after an IPO) because the IPO was too recent or because there was a delay in data availability (further described in Appendix A). Our dataset distinctly outnumbers previous studies: 195 observations in Jewartowski and Lizińska (2012), 211 in Jelic and Briston (2003), and 103 in Lyn and Zychowicz (2003). Noteworthy, Jewartowski and Lizińska (2012) explicitly exclude delisted companies in their research of IPO underpricing effects. We argue that such treatment introduces severe survivorship bias into research on IPO underpricing. In our dataset, there are 178 companies with a market status other than 'currently active', including 58 delisted companies in the WSE and 120 delisted companies in the NC. We include IPO data and subsequent return of the delisted companies in all of our analyses to mitigate the survivorship

¹⁵ In Figures 1-3, as well as in the expanded time-series and cross-sectional statistics for the first-day returns reported in Tables B2 and B3 in Appendix B respectively, we include more data for descriptive purposes. This data goes back to January 2005 and predates the inception of the NC market. The total sample therein includes 874 IPOs: 387 listings in the WSE and 469 listings in the NC. This data starts in 2005 because (1) the stock exchange does not provide a comprehensive IPO list before 2005 and (2) prospectuses for earlier IPOs are often not available.

bias. Otherwise, the results would be biased towards the more successful IPOs. In rare cases of re-listings, we use the data of the chronologically first IPO. We are also the only scholars thus far to include observations from the NC market for analysing IPO underpricing effects. We point to the advantages of our dataset not to claim the superiority of our results over previous research. Instead, we strive to establish a viable reference point to communicate the reliability of our dataset.

We winsorize the data at 0.5% and 99.5% of observations to limit the influence of outliers. We examine winsorization choices on a case-by-case basis and test the sensitivity of our results (see Leone, Minutti-Meza, & Wasley, 2019; Wilson, 1997). We do not winsorize returns due to negative skewness, but we do winsorize IPO size to reduce the impact of the extreme outliers. Table C5 in Appendix C reports our main results corresponding to three different winsorization choices: 0% (no winsorization), 0.5% (winsorizing at 0.5% and 99.5%) and 1% (winsorizing at 1% and 99%). The sensitivity analysis confirms that our results are robust to the alternative winsorization choices.

5.2. IPO characteristics and IPO distribution by sector and in time

Table 1 presents descriptive statistics for the main variables, separately for the sample of companies listing in the WSE and the NC. For the period of January 1, 2005, to June 30, 2017, we report an average IPO size of 177 million PLN in the WSE and 2.6 million PLN in the NC. Table C1 in Appendix C reports the expanded summary statistics with additional percentiles of the empirical distributions of our main variables.

Table 2 Panel A summarises the sample distribution across economic sectors, and Panel B—in time, separately for listings in the WSE and the NC. We classify firms into industries using Global Industry Standard Classification ('GICS').¹⁶ We observe no material differences in sector distribution between the two markets under research. In Panel B, we see an evident drop in the number of IPOs in both markets in the aftermath of the global financial crisis, particularly in the year 2009. The highest total number of IPOs (184) appears in 2011. During that period, the European Union targeted significant subsidies towards small and medium enterprises in Poland. Part of this support program was intended to finance equity issuances in the NC and covered up to half of the total IPO cost for the participating companies. Because the subsidy

¹⁶ Bhojraj et al. (2003) compare alternative industry classifications (including SIC, NAICS, or Fama and French classifications) and argue that GICS outperforms other taxonomies in identifying comparable companies.

program ended in 2011, most firms went public by the end of the year, explaining an increased number of IPOs in that year and a subsequent decline.¹⁷

[Insert Table 1. here]

5.3. IPO underpricing and long-term underperformance

Figure 1 presents unadjusted first-day returns of Polish IPOs over time. The visual inspection of the plot reveals that most of the sample IPOs yield positive first-day returns. Further, the figure indicates that except for the year 2007, the NC first-day returns are consistently higher than the WSE returns. NC sample is also associated with the higher dispersion of the IPO returns. Moreover, there is a noticeable drop in the number of IPOs and an increase in the dispersion of returns during the global financial crisis, particularly in the year 2009. The highest number of IPOs is observed in the post-crisis years 2010 and 2011, which is consistent with the descriptive evidence.

[Insert Figure 1. here]

Table 1 presents empirical estimates for the IPO underpricing and long-term underperformance of equities listed in the two Polish stock markets. The equally-weighted average of the unadjusted first-day return for WSE (NC) is approximately 5.8% (40.0%), indicating a high degree of underpricing, especially in the NC market. The dispersion of NC first-day returns varies from -90.6% to as much as 2,900%. The corresponding minimum and maximum values for WSE first-day returns are -74.07% and 242.0%, respectively. This evidence aligns with Miller's (1977) theory of divergence of opinions among investors and reaffirms the inherent characteristics of the NC, a stock market dedicated to innovative, technology-related and less mature companies (Jackowicz et al., 2017).

The use of IPO size weights does not materially affect the interpretation of the previous results: the average first-day returns for WSE and NC equal to 5.5% and 25.4%, respectively. Given that equally-weighted returns are higher than size-weighted returns and that NC returns are higher than WSE returns, our evidence is consistent with the evidence of smaller IPOs

¹⁷ We also look at the industry composition over time. To the degree that industry membership proxies for significant differences in firm characteristics, this is informative about whether important time trends in the IPO characteristics drive our results. Upon investigation, we do not see significant changes in the industry composition of our IPO sample. We report the industry composition over time in Appendix C Figure C1. One exception is that we see the decline in the number of IPOs in the consumer discretionary sector. Nevertheless, our regression results are robust to the exclusion of the IPOs from that sector. We report an additional robustness check using the alternative sample that excludes IPOs from that sector in Appendix C Table C7.

generating relatively higher IPO returns. This result is also in accordance with the idea of a size premium.

[Insert Table 2. here]

In line with empirical evidence from the U.S. market (e.g., Ritter, 1991), as well as from the foreign developed and emerging markets (e.g., Boulton, Smart, & Zutter, 2011) cumulative market-adjusted returns for newly floated stocks drift downwards after the initial first-day leap. In the Polish stock markets, average unadjusted returns also drop in the initial period following an IPO, which is consistent with Miller (1977). Comparing the average returns for 30 days following an IPO, we observe a downward trend in stocks' performance: an average WSE issue (in an equally weighted portfolio) delivers a 1.8% return, whereas an average NC issue delivers a 43.1% return. Therefore, compared to first-day returns of 5.8% and 40.0% for the WSE and the NC, respectively, an average stock accrues an incremental return of -4.0 percentage point and 3.1 percentage point respectively over the period from day 2 to day 30 after an IPO. Once we consider the IPO-size-weighted returns, stocks in both markets yield lower returns than over the first day of listing only: 4.9% and 21.6% in the WSE and NC, respectively.

Figures 2 and 3 show the average short- and long-term performance of the WSE and NC IPOs. Figure 2 presents the unadjusted, IPO-size-weighted (Panel A) and equally weighted (Panel B) returns accumulated until the 360th day after an IPO. The NC IPO-size-weighted stock returns drift downwards immediately after the upsurge on the first day. Their WSE counterparts do not follow that exact path: the WSE IPO-size-weighted stock returns rise gradually throughout the period, with a slight indication that they are sloping downwards around day 330. The NC stock returns experience numerous erratic movements, with frequent changes from high to low. Such variability in returns may be related to the specific characteristics (i.e. immaturity) of stocks listed in this market, especially when compared to the WSE companies.

[Insert Figure 2. here]

Next, we look at the market-adjusted returns. Figure 3 plots IPO-size-weighted (Panel A) and equally-weighted (Panel B) WSE and NC returns accumulated until the 360th day after an IPO. Despite numerous fluctuations, the returns indisputably descend over time, confirming previous findings. Further, NC portfolio returns are considerably more volatile than WSE portfolio returns (Panel A). This observation is consistent with differences in the risk-return profiles of WSE and NC stocks, and is further confirmed by lower standard deviations of the WSE returns

compared to their NC counterparts (see Table 2). The cumulative market-unadjusted equallyweighted returns indicate flat to slightly increasing trends (Panel B).

[Insert Figure 3. here]

We also conduct a cross-section and time-series analysis of the first-day returns. Partitioning the sample by years and industries enables a more thorough understanding of the IPO underpricing effect, which does not seem to be driven by the year or industry under study. Time-series analysis indicates some cyclicality in the first-day returns. Yet, the relationship is not completely clear as the relative consequences of the global financial crisis were rather mild for the Polish financial sector. Cross-section analysis of the first-day returns reveals considerable cross-sectional variation in the IPO returns. Ultimately, the estimates for the decomposed returns reinforce earlier evidence of the IPO underpricing effect (Stoll & Curley, 1970). We provide the detailed results and more thorough analysis thereof in Appendix B, Tables B1 and B2 for the time-series and cross-section of first-day returns, respectively.

6. Empirical Results

Figure 4 adds to the practical motivation to our identification strategy and predictions. We compare the cumulated percentage changes in the aggregated pension funds' AUM, in the equity part of the aggregated pension funds' AUM and the Warsaw Stock Exchange WIG index. The cumulated percentage changes are scaled relative to the yearend of 2013.

[Insert Figure 4. here]

In 2014, we observe a material drop in the AUM due to the one-off transfer of funds from the second to the first pillar of the Polish pension system. However, this transfer encompassed only government-guaranteed debt securities and cash. A more critical observation is what happened with pension funds' equity investments. Before 2014, pension funds' equity holdings have been decreasing less and increasing more than the WIG index. It is due to the internal benchmark mechanism that funds have been following each other's equity investments rather than pursuing active returns, which Voronkova and Bohl (2005) classify as herding and provide evidence thereof in the Warsaw Stock Exchange. Consequently, pension funds' performance has been mirroring the WIG index is to be attributed to the positive and high net cash inflows of contributions (which funds have been investing in equities listed in the WSE). Interestingly,

pension funds' equity holdings have not been surpassing the WIG index after February 2014. Instead, changes in pension funds' equity holdings have been strictly following the WIG changes. This is suggestive of not much additional net capital inflows, and non-increasing equity investments. If that is true that changes in equity holdings are only due to changes in assets prices rather than to incremental investments, this observation might also indicate the continuation of herding behaviour among funds when it comes to their equity engagement. We are inclined to think that the regulatory change of June 2014 related to departing from the internal benchmark mechanism as a measure of funds' performance did not have a considerable impact on their investment strategies. Providing empirical evidence thereto is, however, out of the scope of the current paper. Conversely, the above observations help us develop a prior that once there is no more incentive for herding, there will be less pressure for higher IPO valuations ('pension premium' in IPO proceeds) because not all pension funds would invest in IPOs that other pension find attractive.

Based on the casual observation of Figure 4, from the year 2014 onwards we expect lower aggregated equity investments in newly listed stocks, which should negatively impact IPO proceeds (IPO size). Since it is difficult to make a precise prediction upon graphical illustration only, the effect of the government-forced changes on the IPO pricing and performance becomes the focus of further analysis.

6.1. Regulation-Induced Changes in the IPO Market

We begin by testing for the differences in means between the pre- and post-treatment subsamples. We split the sample by the IPO date, using February 3, 2014 as the cut-off date. We run a two-sample *t*-test on the IPO first-day returns, on the return proxies for the short- and long-term underperformance, and on the IPO size. We perform a separate test for WSE and NC new listings.

[Insert Table 3. here]

The regulatory changes affected both the size of pension funds AUM and the way they invest capital (investment policy). Therefore, we are expecting this treatment to impact IPO pricing and the subsequent performance of the newly listed stocks. Table 3 posits that there is an economically and statistically significant difference in the mean IPO size between the pre-and post-treatment subsamples in the WSE. This finding does not depend on whether we use short-or long-time window implementation strategy. In both cases, the pre- and post-treatment mean IPO size in the WSE sample decreases materially, i.e. by 142 million PLN when using a short-

time window, and by 128 million PLN when using long-time window. Both estimates are significant at 5% level. In terms of economic magnitude, the post-intervention decline corresponds to around 70% of the pre-intervention mean IPO size. At the same time, the difference in the pre- and post-treatments mean IPO size in the NC market lacks statistical and economic significance. These results provide first empirical evidence of lower IPO valuations associated with the regulatory change in the stock market with a strong presence of pension funds. The effect is otherwise unobservable (in the NC subsample).

We fail to uphold similar conclusions for the differences in the mean first-day IPO returns and the mean short- and long-term underperformance (cumulated IPO returns) in either of the stock markets. We conclude that the regulatory change did not affect the degree of IPO underpricing. The null result is consistent with the signalling model (Ritter & Welch, 2002). If the underpricing is understood as a tool to signal quality to investors, there is no reason to believe that the demand shock on the investors' side would affect underwriters' decision to underprice to a different degree. This result is also in line with practitioners' views. According to the fund managers consulted in our research process, pension funds actively invest at IPOs but do not tend to rebalance portfolios in the short-term. The latter insight means that they hardly ever trade stock on the first day after an IPO nor change their positions during the 360 days after an IPO.

Incidentally, we note that the negative demand shock on the investors' side could also affect overall market performance, especially the IPO risk-adjusted long-term performance. In Table C3 in Appendix C, we test for the difference between pre- and post-treatment sample average unadjusted returns in the WSE that are independent of the market performance. This difference shows as insignificant, regardless of the time horizon used. Therefore, we discard the possibility that the results for market-adjusted returns reported in Table 3 are driven by a combination of the offsetting effects on IPO returns and overall market performance. In Table C4 in Appendix C, we also test whether our conclusion is driven by inflated IPO returns in the years of the global financial crisis and we report that excluding years 2008 and 2009 from our sample does not have any effect on the degree of IPO underpricing. Table C8 in Appendix C reports our regression results in the subsample that excludes years 2008 and 2010 of the global financial crisis.

Figure 5 shows the visualization of our result. The figure plots mean annual IPO size for the WSE (Panel A) and NC (Panel B) markets in 3 years preceding and following the event (parallel trend lines). As it transpires, there was a significant drop in the mean IPO size following the

regulatory change in early 2014. In fact, in the years 2014, 2015 and 2016, the WSE IPO valuations were consistently at their lowest levels in terms of the average offering size. This is not the case in the NC sample, what validates our expectations. This finding also eliminates potential concerns that the changes in IPO pricing may result from unobservable events or other changes in macroeconomic and market conditions.

[Insert Figure 5. here]

Next, we show the results of the nonparametric bootstrap test. We aim to formally assess the statistical significance of our finding without making any distributional assumptions. Figure 6 shows the distribution of bootstrapped pre-post differences in the mean IPO size for the WSE (Panel A) and NC (Panel B) market. In bootstrapping tests, we sample IPOs with replacement and calculate the difference between the mean IPO size in the pre-treatment period and the mean IPO size in the post-treatment period. We use February 3, 2014 as our cut-off date and symmetric 3-year period around the cut-off in the IPO sampling procedure (the short-time window). We plot the density based on 10,000 sampling iterations. The value of sample the difference in our original sample is illustrated with the solid vertical line. Figure 6, Panel A (WSE) shows the original sample difference of -142.04 million PLN. We argue that this is not driven by chance (is statistically significant). In fact, our observed original sample difference of -142.04 million PLN corresponds to the t-statistic of 2.07, as only 1.9% observations lie below the observed mean difference. In contrast, Figure 6, Panel B (NC) reports the original sample difference of -0.31 million PLN, which is no different from the mean difference resulting from the bootstrapping. In this case, the t-statistic equals to 0.34, as 37.0% of observations lie below the observed mean difference. We repeat the same analysis using full data from January 1, 2008 to June 30, 2017 (the long-time window). This implementation strategy yields similar results. Therefore, based on the random sampling testing procedure, we confirm a statistically and economically significant decline in the IPO size in the sample where pension funds are heavily invested (WSE), but not in the sample that is independent of changes in pension funds investment policies (NC). We also run a permutations test to arrive at similar conclusions. We show the distribution of permutated pre-post differences in the mean IPO size for the WSE and NC market in Figure C2 in Appendix C. We additionally provide a summary of all nonparametric test estimates in Table C2 in Appendix C.

[Insert Figure 6. here]

6.2. Event Study

We implement the difference-in-differences estimator to test for the impact of the 2014 regulatory change to the pension funds' AUM and investment policy on the IPO pricing and stock's subsequent performance. We estimate the average response of the IPO size, market-adjusted first-day returns, market-adjusted short- (30-days) and long-term (360-days) underperformance to the regulatory change on the demand side. As per difference-in-differences specification, we use the WSE IPOs as the treated group and the NC IPOs as the control group. Table 4 shows the estimates of the average dependent variables' response to the exogenous shock based on the model as per Equation 6. To control for unobservable industry-level and year heterogeneity, we use industry and year fixed effects in all regressions.

[Insert Table 4. here]

In Table 4, we report material and statistically significant change in the IPO size (winsorized at 0.5% and 99.5%). Our results indicate that the average IPO size in the treated group went down by 128 million PLN (almost 41 million USD¹⁸) after the regulatory change. The *p*-value of the estimate is 5.9%. The estimate increases to 152 million PLN if we use unwinsorized data (Table C5 in Appendix C). These results are reported for the short-time window implementation strategy. When we implement a long-time window strategy, our findings prevail. The point estimate for the effect of the regulatory change on IPO size is -107 million PLN, significant at the 5% level (p-value of 4.9%). Furthermore, irrespective of the implementation strategy, the statistical significance of the WSE estimate at the 1% level validates the difference in crossmarket characteristics: IPOs in the WSE are on average 76 times bigger in their valuations than their NC counterparts (as per descriptive statistics in Table 1). Against this backdrop, we argue that government-forced changes to pension funds AUM and investment policy of 2014 resulted in lower IPO proceeds. We also provide first robust empirical evidence in support of the practitioners' claim that the described regulatory changes suppressed the level of IPO valuations, since historically pension funds have been crucial buyers of shares responsible for the so-called 'pension premium.'

As reported in Table 4, we find some evidence that government-forced changes on the demand side affected IPO first-day returns. Specifically, the difference-in-differences estimate for the first-day risk-adjusted returns shows a decline of 17 percentage point, significant at 5% level.

¹⁸ Exchange rates are as per the official Central Bank of Poland statistics; retrieved on February 23, 2019, from <u>https://www.nbp.pl/homen.aspx?navid=archen&c=/ascx/TabArchAen.ascx&n=14a022en</u> for February 3, 2014.

This decline is consistent with the book building literature which argues that IPO underpricing is a mean of compensating institutional investors for revealing value-relevant information during the book building process (Aggarwal et al., 2002; Benveniste & Spindt, 1989; Chemmanur et al., 2010). Furthermore, we see no impact on the long-term IPO performance, which is consistent with the fact that the association between institutional trading and long-term IPO performance decays over time (Chemmanur et al., 2010).

For robustness purposes, we additionally test whether our results for the effect of regulatory change on the IPO short- and long-term underperformance could have been biased by improper treatment of IPO observations that listed during the pre-treatment period but which 30-day and 360-day interval for aftermarket performance measurement are also in the range of the post-treatment period. In other words, we investigate whether the short-term performance of stocks that debuted between January 3, 2014 and February 2, 2014, and long-term performance of stocks that debuted between February 3, 2013 and February 2, 2014 capture some impact of the policy change. To test this, we adjust the threshold so that the post-intervention period includes all the returns that could have been potentially affected. The results remain unchanged. We report these results in Table C9 in Appendix C.

In Table 5 we report the results from estimating the difference-in-differences regression that includes additional controls for firm characteristics. We control for size, profitability, and VC-backing status, as described earlier.

[Insert Table 5 here]

The results presented in Table 5 confirm our main results. Specifically, we report material and statistically significant change in the IPO size (winsorized at 0.5% and 99.5%). Our results indicate that the average IPO size in the treated group decreased by 96 million PLN. In terms of economic magnitude, this estimate is smaller than the estimate in the regression that omits controls (128 million PLN) but remains statistically significant. This result supports our conclusion that the government-forced changes to pension funds AUM and investment policy of 2014 resulted in lower IPO proceeds.

An alternative (and simpler) specification of our model is when instead of the year fixed effects, we include the post-treatment indicator, $Post_i$. Our results are robust to this alternative specification (see Table C6 and Table C11 in Appendix C for the results in the models without and with the control variables, respectively). Furthermore, explicitly including the $Post_i$ variable allows us to make linear predictions. As per table C6 in Appendix C, we report the

overall decrease in the IPO size equal to 132 million PLN and the decrease in the degree of the first-day underpricing of 11 percentage points in the WSE. Although pension funds hold only 12% of the equity in the WSE, their effect on IPO pricing is undoubtedly material.

We next supplement our analysis with a cross-sectional test based on the IPO size. Zalewska (2006) documents a blue-chip bias among Polish pension funds, whereby pension funds prioritize issuers with the largest capitalization in their equity portfolios.¹⁹ Since pension funds' investments are concentrated around the local blue-chips, we expect our results to be particularly evident in the subsample that includes the largest IPOs. We motivate our prior with the fact that the impact of the regulatory change was strongest for the firms with the highest pension funds' participation. Accordingly, each year we sort all IPOs based on the quartiles of IPO size, separately for the treatment and control sample.²⁰ We then rerun our difference-in-differences regressions across the four size partitions. We report the results in Table 6. For brevity, we only show the coefficients on the interaction term (*WSE* × *Post*), which is our main coefficient of interest.

[Insert Table 6. here]

The table provides strong support for our hypothesis. Larger issuers drive the negative impact on the IPO size. The difference-in-differences estimates are statistically significant for the two biggest partitions (quartile 3 and 4), but insignificant for the smallest two partitions (quartile 1 and 2).²¹ Specifically, using the short-window test, the decline is 24.5 million PLN in the third size quartile, and 669.3 million PLN in the fourth quartile, both estimates statistically significant. The decrease in the largest size quantile remains significant in the long-time window test. This result is consistent with our prediction that the decline in the IPO size that we observe in the full WSE IPOs sample is a result of the 2014 regulatory treatment of the Polish pension funds, which invest primarily in the larger companies (Field & Lowry, 2009).

Furthermore, Table 6 reveals a statistically significant decline in the IPO first-day returns following the regulation in the largest-IPO-size partition. Specifically, the average IPO first-

¹⁹ The large-cap bias is not confined to the Polish stock market only. It is also a significant regulatory challenge in the developed markets (see Fleming, 2017).

²⁰ By sorting stocks separately for WSE and NC samples our quantile test compares the IPOs based on the relative size. To make sure that the differences in firm size between treatment and control firms in the given portfolios are not driving this result, we additionally rerun this analysis using an alternative specification that controls for firm size, in addition to profitability and VC backing status. We report those results in Table C12 in Appendix C.

²¹ We also report the coefficients on the log size, as the distribution of IPO size is less asymmetric in the narrowlydefined size partitions. Otherwise, the log transformation introduces the attenuation bias. This is because the larger IPOs have higher Pension funds ownership and hence contribute the most to our results.

day returns decreased by 30.0 percentage points when we used a short-term window implementation strategy, and 24.1 percentage points when we used a long-term window, both statistically significant. This decline is consistent with the book building literature (e.g., Benveniste & Spindt, 1989), which posits that IPO underpricing is a mean of compensating institutional investors for revealing value-relevant information during the book building process (Chemmanur et al., 2010). If there is a positive association between the institutional investors' participation in the IPO book building process and the level of IPO underpricing, then an exogenous shock to institutional investors' participation, that creates a negative demand shock, results in a lower degree of IPO underpricing.

Then, we report weak evidence of a negative impact of the 2014 reform on the IPO long-term underperformance. In the largest-IPO-size partition, there is a statistically significant decline in the 30-day market returns of 69.4 percentage point when using the short-time window and of 63.1 percentage point when using the long-time window. This result is consistent with the information asymmetry literature (Field & Lowry, 2009), which suggests a positive association between institutional trading and IPO long-term performance. Also, there are no significant effects in the 360-day post-IPO period. This is in line with the stylized fact that the association between institutional trading and IPO long-term performance decays over time (Chemmanur et al., 2010).

Lastly, we discard the possibility of more firms deciding to stay private after the reform. First, we do not observe a reduction in the number of IPOs after the reform (in contrast to the U.S. market as reported by Gao, Ritter, & Zhu, 2013). As per Table 1, we report that in the years following the 2014 reform, IPO activity proxied by the number of IPOs did not decrease as compared to the years preceding the reform. We also look at the number of IPO withdrawals for each year in our time-series. Based on the Bloomberg data, the withdrawal rate was 11% in 2013 (the year before the reform), 3.4% in 2014 (event year). Between 2005 and 2013, an average yearly withdrawal rate was 10%, while between 2014 and 2017, it was 3% on average. In our study, we also report lower underpricing in the period after the pension reform in contrast to Dorsman and Gaunopoulos' (2013) findings. Against this backdrop, we conclude that it is unlikely that the pension reform resulted in fewer companies going publish with equity in the stock market of Poland. Conversely, we argue that after the reform market participants constituted a 'new market equilibrium', where listing companies accept new market conditions (i.e. the fact that pension funds have less money to invest) and they go on with an IPO despite the realization that their IPO proceeds would be lower than if they listed before 2014.

6.3. Robustness

Finally, we examine whether our results are robust with regard to the date of the regulatory change. Specifically, we would like to disprove an argument that IPO size changes multiple times in our sample period, and the regulatory change only happened to be one of such events. We are also interested whether the precise date of the event is the very moment of changes in IPO proceeds. The alternative view is that changes in the IPO pricing might have been occurring gradually in anticipation of the new regulation (pension funds demand for new equities decreased already when there were rumours about the oncoming changes to the Polish pension system). Figure 7 plots the *p*-values of the difference-in-differences coefficient in the regression where the IPO size is the response variable, relative to the alternating cut-off date. A visual inspection enables us to validate that February 2014 was indeed a critical timing for material changes in the average IPO size in the WSE treated group. Only at the beginning of the year 2014, the *p*-value of the difference-in-differences coefficient falls to the close-to-zero level, which means that the estimate becomes statistically significant. This result is particularly pronounced when we use the symmetric bandwidth of 3 years around the threshold (short-time window implementation strategy). Moreover, the plot indicates that the result is relatively insensitive to the exact cut-off date: the difference-in-differences estimate is statistically significant at the 5% level as long as we choose a date around the implementation of the regulatory change. We argue that this is driven by the nature of our data (IPOs do not happen every day). A lack of observations for each trading day, while running placebo testing using each trading day as the cut-off, results in the *p*-values being significant for the period around the event rather than for an individual point in time. However, there is no trait of IPO size being lower in the period preceding the regulatory change when it was being prepared and consulted. Hereto, we do not report statistically significant *p*-value estimates for the period of one year preceding the event. Further, we also reject the argument that the event might have been expected by companies that hastened the IPO process in anticipation of the law change to avoid lower IPO proceeds. In the data we do not see a tendency to go public earlier to avoid declining demand. The number of IPOs in the WSE market in 2014 is higher than its counterpart in 2013 (see Table 1 for the number of IPOs each year). The placebo also addresses the concerns that our results are directly attributable to the Global Financial Crisis because most of the simulated windows are contained entirely in the post-crisis period.

[Insert Figure 7. here]

Accordingly, we argue that the estimates for the causal effect of regulatory changes on the IPO size presented in Table 4 are robust to the placebo testing related to the event date.

7. Conclusion

The Polish pension system was established in 1999, and in time proved unsustainable. In early 2014, a pension fund reform was implemented to solve the country's fiscal problems. The main changes of the 2014 reform encompassed lower contributions to the pension plans managed by open (private) funds, followed by the removal of the internal benchmark mechanism as a measure of pension fund's performance. These changes had a direct effect on pension funds AUM and their investment policy. In this paper, we go one step further and investigate the indirect effects of the 2014 regulatory change on capital markets, and specifically on the IPO pricing and stock's subsequent performance.

Using a novel data sample, we provide robust evidence for the IPO underpricing effects in the two equity markets in Poland. We find that between 2005 and 2017 an average WSE (NC) IPO yields a 10.59% (37.56%) market-adjusted first-day return when equal weights are used and 8.26% (24.1%) first-day market-adjusted return when IPO-size weights are used. Significant differences in returns between equally- and IPO-size-weighted portfolios indicate the superior performance of small IPOs. The estimates for short- and long-term performance of stocks listed in both equity markets (cumulated over 30 and 360 days after an IPO) confirm the long-term underperformance effect. The average 360-day market-adjusted return for an IPO debuting in the WSE (NC) is 11.61% (6.49%) when equally-weighted, and -0.31% (-10.53%) when IPO-size-weighted. Notably, the test for long-term IPO underperformance is dependent upon the risk adjustment of the returns.

The unique opportunity to exploit a quasi-natural experiment arising from the Polish government's regulatory treatment of pension funds enables us to shed new light on what factors influence new issues' market. We show that the decreasing net cash inflows to pension funds' AUM and abandoning internal benchmark as a measure of funds' performance result in less demand for new equities that manifests itself in significantly smaller IPO size. In the reform aftermath, firms that list in the WSE market, raise substantially less money in the process: 142 million PLN less on average using a simple *t*-test approach and estimating the difference between the pre- and post-treatment mean IPO size. Regression results reinforce our conclusions and confirm the causal effect—the incremental impact of the regulatory-forced

changes to pension funds AUM and investment policy on the IPO size is estimated at -107 million PLN and is statistically significant. This result is robust to using different bandwidth around the cut-off date placebo testing. There is no such effect in the NC market, where pension funds are not invested. We leave the question of whether the impact of the reform on valuations reverses in the long-term as a potential avenue for further research.

Testing for the differences in the IPO underpricing effects before and after the governmentforced changes to pension funds' AUM and investment policy reveals that an exogenous shock had some impact on the degree of underpricing. The effect was especially among larger firms, where pension funds are primarily invested. We report no impact of the reform on the degree of IPO long-term underperformance. Our findings contribute to the literature on the drivers of IPO underpricing. We report a significant decrease in the IPO underpricing for the subsample of issuers where the Polish pension funds invest the most. This decline is consistent with the rich book building literature which states that IPO underpricing is a mean of compensating institutional investors for revealing value-relevant information during the book building process (Baron, 1982; Beatty & Ritter, 1986; Benveniste & Spindt, 1989; Chemmanur et al., 2010; Rock, 1986).

We can interpret our results in a negative tone, that is after the regulatory change companies raise less money in the primary stock market, or in a positive tone, that is the reform eliminated an artificial 'pension premium' of higher valuations, and thus contributed to market efficiency. In any case, reforming pension system for the sake of solving fiscal problems turned out to have spillover effects on capital markets. Specifically, a government trying to fix the unsustainable pension system by limiting net cash inflows to the privately-managed pension funds directly contributed to a decrease in the average IPO size. By providing empirical evidence thereof, we deliver novel insights into the existing IPO literature.

This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

References

- Aggarwal, R., Prabhala, N. R., & Puri, M. (2002). Institutional Allocation in Initial Public Offerings: Empirical Evidence. *The Journal of Finance*, 57(3), 1421–1442. https://doi.org/10.1111/1540-6261.00465
- Barber, B. M., & Lyon, J. D. (1997). Detecting long-run abnormal stock returns: The empirical power and specification of test statistics. *Journal of Financial Economics*, 43(3), 341–372. https://doi.org/10.1016/S0304-405X(96)00890-2
- Baron, D. P. (1982). A Model of the Demand for Investment Banking Advising and Distribution Services for New Issues. *The Journal of Finance*, *37*(4), 955. https://doi.org/10.2307/2327760
- Barth, M. E., Landsman, W. R., & Taylor, D. J. (2017). The JOBS act and information uncertainty in IPO firms. *The Accounting Review*, 92(6), 25–47. https://doi.org/10.2308/accr-51721
- Beatty, R. P., & Ritter, J. R. (1986). Investment banking, reputation, and the underpricing of initial public offerings. *Journal of Financial Economics*, *15*(1–2), 213–232. https://doi.org/10.1016/0304-405X(86)90055-3
- Benveniste, L. M., & Spindt, P. A. (1989). How investment bankers determine the offer price and allocation of new issues. *Journal of Financial Economics*, 24(2), 343–361. https://doi.org/10.1016/0304-405X(89)90051-2
- Bertrand, M., Duflo, E., & Mullainathan, S. (2004). How Much Should We Trust Differences-In-Differences Estimates? *The Quarterly Journal of Economics*, 119(1), 249–275. https://doi.org/10.1162/003355304772839588
- Bhojraj, S., Lee, C. M. C., & Oler, D. K. (2003). What's My Line? A Comparison of Industry Classification Schemes for Capital Market Research. *Journal of Accounting Research*, 41(5), 745–774. https://doi.org/10.1046/j.1475-679X.2003.00122.x
- Blake, D., & Timmermann, A. (2002). Performance Benchmarks for Institutional Investors: Measuring, Monitoring and Modifying Investment Behavior. In J. Knight & S. Satchell (Eds.), *Performance Measurement in Finance: Firms, Funds and Managers* (pp. 108–141). Oxford: Butterworth Heinemann. https://doi.org/10.1016/b978-075065026-7.50006-7
- Boulton, T. J., Smart, S. B., & Zutter, C. J. (2011). Earnings Quality and International IPO Underpricing. *The Accounting Review*, 86, 483–505. https://doi.org/10.2307/29780243
- Brau, J. C., & Fawcett, S. E. (2006). Initial Public Offerings: An Analysis of Theory and Practice. *The Journal of Finance*, *61*(1), 399–436. https://doi.org/10.1111/j.1540-6261.2006.00840.x
- Chang, C., Chiang, Y.-M., Qian, Y., & Ritter, J. R. (2016). Pre-market Trading and IPO Pricing. *The Review of Financial Studies*, *30*(3), 835–865. https://doi.org/10.1093/rfs/hhw032
- Chemmanur, T. J., Hu, G., & Huang, J. (2010). The Role of Institutional Investors in Initial Public Offerings. *Review of Financial Studies*, 23(12), 4496–4540. https://doi.org/10.1093/rfs/hhq109
- Cienski, J. (2014). Poland pensions law signed but is likely to face legal challenge | Financial Times.
- Coval, J. D., & Stafford, E. (2007). Asset fire sales (and purchases) in equity markets. *Journal of Financial Economics*, 86(2), 479–512. Retrieved from http://econpapers.repec.org/article/eeejfinec/v_3a86_3ay_3a2007_3ai_3a2_3ap_3a479-512.htm
- Damodaran, A. (2018). Damodaran Online. Retrieved July 1, 2018, from http://pages.stern.nyu.edu/~adamodar/New_Home_Page/datacurrent.html#discrate
- Davis, E. P. (1998). Pension Funds. Retirement-Income Security and Capital Markets: An International Perspective. Oxford University Press. https://doi.org/10.1093/acprof:oso/9780198293040.001.0001
- Dorsman, A., & Gounopoulos, D. (2013). European Sovereign Debt Crisis and the performance of Dutch IPOs. *International Review of Financial Analysis*, *30*, 308–319. https://doi.org/10.1016/j.irfa.2013.07.003
- Ellis, K., Michaely, R., & O'Hara, M. (2000). When the Underwriter Is the Market Maker: An Examination of Trading in the IPO Aftermarket. *The Journal of Finance*, 55(3), 1039–1074. https://doi.org/10.1111/0022-1082.00240
- Fama, E. F., & French, K. R. (2015). A five-factor asset pricing model. *Journal of Financial Economics*, *116*(1), 1–22. Retrieved from

http://econpapers.repec.org/article/eeejfinec/v_3a116_3ay_3a2015_3ai_3a1_3ap_3a1-22.htm

- Field, L. C., & Lowry, M. (2009). Institutional versus Individual Investment in IPOs: The Importance of Firm Fundamentals. *Journal of Financial and Quantitative Analysis*, 44(3), 489–516. https://doi.org/10.1017/S0022109009990032
- Fleming, R. A. (2017). Enhancing the Demand for IPOs. NASAA 2017 Public Policy Conference. Washington, DC: NASAA 2017 Public Policy Conference. Retrieved from https://www.sec.gov/news/speech/flemingenhancing-demand-ipos-050917
- FTSE. (2017). *Press release from 29 September 2017*. Retrieved from https://www.ftserussell.com/sites/default/files/press_releases/ftse_annual_country_classification_review_s eptember_2017_final_0.pdf
- Gao, X., Ritter, J. R., & Zhu, Z. (2013). Where have all the IPOs gone? *Journal of Financial and Quantitative Analysis*. Cambridge University Press. https://doi.org/10.1017/S0022109014000015
- Góra, M. (2014). Redesigning pension systems. https://doi.org/10.15185/izawol.51
- GPW. (2017). WSE Statistic Bulletin. First half-year 2017. Retrieved October 18, 2018, from https://www.gpw.pl/gpw-statistics#4
- GPW. (2018). GPW Main Market GPW key statistics. Retrieved from https://www.gpw.pl/gpw-key-statistics
- Hausman, C., & Rapson, D. S. (2018). Regression Discontinuity in Time: Considerations for Empirical Applications. Annual Review of Resource Economics, 10(1), 533–552. https://doi.org/10.1146/annurevresource-121517-033306
- Holzmann, R. (2000). The World Bank Approach to Pension Reform. *International Social Security Review*, 53(1), 11–34. https://doi.org/10.1111/1468-246X.00061
- Ibbotson, R. G., & Jaffe, J. F. (1975). "Hot issue" markets. *The Journal of Finance*, *30*(4), 1027–1042. https://doi.org/10.1111/j.1540-6261.1975.tb01019.x
- IMF. (2014). Republic of Poland. Selected Issues. IMF Country Report, No. 14/174.
- Jackowicz, K., Kowalewski, O., Kozłowski, Ł., & Roszkowska, P. (2017). Issuing bonds, shares or staying private? Determinants of going public in an emerging economy. *Post-Communist Economies*, 29(1), 1–26. https://doi.org/10.1080/14631377.2016.1226771
- Jelic, R., & Briston, R. (2003). Privatisation Initial Public Offerings: the Polish Experience. European Financial Management, 9(4), 457–484. https://doi.org/10.1111/1468-036X.00232
- Jewartowski, T., & Lizińska, J. (2012). Short- and Long-Term Performance of Polish IPOs. *Emerging Markets Finance and Trade*, 48(2), 59–75. https://doi.org/10.2753/REE1540-496X480204
- Journal of Laws 2013, Item 1717 (2013). Republic of Poland.
- KNUiFE. (2000). Security through competition: performance analysis of the second pillar. Warsaw.
- Kominek, Z. (2012). Regulatory induced herding? Evidence from Polish pension funds. *Economic Change and Restructuring*, 45(1–2), 97–119. https://doi.org/10.1007/s10644-011-9111-2
- Kucharczyk, K. (2016). OFE na razie nie stronią od rynku pierwotnego. Retrieved from https://www.parkiet.com/artykul/1458936.html&template=restricted
- Lakonishok, J., Shleifer, A., & Vishny, R. W. (1992). The impact of institutional trading on stock prices. *Journal of Financial Economics*, 32(1), 23–43. https://doi.org/10.1016/0304-405X(92)90023-Q
- Leone, A. J., Minutti-Meza, M., & Wasley, C. E. (2019). Influential Observations and Inference in Accounting Research. *The Accounting Review*, (forthcoming). https://doi.org/10.2308/accr-52396
- Liu, X., & Ritter, J. R. (2010). The Effects of Financial independence. *Review of Financial Studies*, 23(5), 2024–2059. https://doi.org/10.1093/RFS
- Ljungqvist, A., Nanda, V., & Singh, R. (2006). Hot Markets, Investor Sentiment, and IPO Pricing. *The Journal of Business*, 79(4), 1667–1702. https://doi.org/10.1086/503644
- Logue, D. E. (1973). On the Pricing of Unseasoned Equity Issues: 1965-1969. *The Journal of Financial and Quantitative Analysis*, 8(1), 91. https://doi.org/10.2307/2329751
- Loughran, T., & Ritter, J. (2004). Why Has IPO Underpricing Changed over Time? Financial Management,

33(3), 5-37. https://doi.org/10.2307/3666262

- Lowry, M. (2003). Why does IPO volume fluctuate so much? *Journal of Financial Economics*, 67(1), 3–40. https://doi.org/10.1016/S0304-405X(02)00230-1
- Lyn, E. O., & Zychowicz, E. J. (2003). The performance of new equity offerings in Hungary and Poland. *Global Finance Journal*, 14(2), 181–195. https://doi.org/10.1016/S1044-0283(03)00011-5
- Miller, E. M. (1977). Risk, Uncertainty, and Divergence of Opinion. *The Journal of Finance*, 32(4), 1151. https://doi.org/10.2307/2326520
- NewConnect. (2017). NewConnect Statistic Bulletin. First half-year 2017. Retrieved October 19, 2018, from https://newconnect.pl/statystyki-okresowe
- PwC. (2018). *IPO Watch Europe 2017*. Retrieved from https://www.pwc.co.uk/audit-assurance/assets/pdf/ipo-watch-europe-2017-annual-review.pdf
- Ritter, J. R. (1991). The Long-Run Performance of initial Public Offerings. *The Journal of Finance*, 46(1), 3–27. https://doi.org/10.1111/j.1540-6261.1991.tb03743.x
- Ritter, J. R. (2017). Initial public offerings: underpricing. Retrieved from https://site.warrington.ufl.edu/ritter/ipo-data/
- Ritter, J. R. (2019). Jay R. Ritter UF Warrington Faculty. Retrieved May 23, 2019, from https://site.warrington.ufl.edu/ritter/files/2019/04/IPOs2018Statistics-1.pdf
- Ritter, J. R., & Welch, I. (2002). A Review of IPO Activity, Pricing, and Allocations. *The Journal of Finance*, 57(4), 1795–1828. https://doi.org/10.1111/1540-6261.00478
- Roberts, M. R., & Whited, T. M. (2013). Endogeneity in Empirical Corporate Finance. In *Handbook of the Economics of Finance* (Vol. 2, pp. 493–572). Elsevier B.V. https://doi.org/10.1016/B978-0-44-453594-8.00007-0
- Rock, K. (1986). Why new issues are underpriced. *Journal of Financial Economics*, 15(1–2), 187–212. https://doi.org/10.1016/0304-405X(86)90054-1
- Roldos, J. (2004). Pension Reform, Investment Restrictions and Capital Markets. *IMF Policy Discussion Papers*. Retrieved from https://ideas.repec.org/p/imf/imfpdp/04-4.html
- Roll, R. (1977). A critique of the asset pricing theory's tests Part I: On past and potential testability of the theory. *Journal of Financial Economics*, 4(2), 129–176. https://doi.org/10.1016/0304-405X(77)90009-5
- Roszkowska, P., & Langer, L. K. (2016a). (Ab)Normal Returns in an Emerging Stock Market: International Investor Perspective. SSRN Electronic Journal. https://doi.org/10.2139/ssrn.2819698
- Roszkowska, P., & Langer, L. K. (2016b). Counterintuitive Investment Opportunities in the WSE. Evidence from the Field of Asset Pricing. *SSRN Electronic Journal*. https://doi.org/10.2139/ssrn.2819693
- Singh, A. (1996). Pension reform, the stock market, capital formation and economic growth: A critical commentary on the World Bank's proposals. *International Social Security Review*, 49(3), 21–43. https://doi.org/10.1111/j.1468-246X.1996.tb01108.x
- Stoll, H. R., & Curley, A. J. (1970). Small Business and the New Issues Market for Equities. *The Journal of Financial and Quantitative Analysis*, 5(3), 309. https://doi.org/10.2307/2329998
- Voronkova, S., & Bohl, M. T. (2005, September). Institutional traders' behavior in an emerging stock market: Empirical evidence on Polish pension fund investors. *Journal of Business Finance and Accounting*. https://doi.org/10.1111/j.0306-686X.2005.00639.x
- Walker, E., & Lefort, F. (2000). Pension Reform And Capital Markets: Are There Any (Hard) Links? SSRN Electronic Journal. https://doi.org/10.2139/ssrn.239531
- Wilson, C. R. (1997). Distortion effects and extreme observations in empirical research: An analysis of the incremental information content of cash flows. *Accounting & Finance*, 37(2), 163–180. https://doi.org/10.1111/j.1467-629X.1997.tb00319.x
- Zalewska, A. (2006). Is locking domestic funds into the local market beneficial? Evidence from the Polish pension reforms. *Emerging Markets Review*, 7(4), 339–360. https://doi.org/10.1016/j.ememar.2006.09.007

Figures and Tables

Figure 1. Unadjusted first-day returns of WSE and NC IPOs from January 2005 to June 2017.

The figure plots the first-day return of all IPOs listed on WSE and NC in the period from January 2005 to June 2017. The first-day return is calculated as the difference between the closing price of the first day of trading and the IPO offer price, divided by that closing price. For presentation purposes, we exclude 7 observations that have the first-day returns exceeding 300%.



Exchange: • WSE 🔺 NC

Figure 2. Unadjusted short- and long-term performance of WSE and NC IPOs from January 2005 to June 2017.

Panels A and B show the performance of stocks following an IPO. Returns are cumulated over the relevant period using the CAR procedure. The solid line depicts the performance of the WSE IPO portfolio, while the dotted line represents the performance of the NC IPO portfolio. Portfolios are formed using 99% winsorized IPO size weights (Panel A) or equal weights (Panel B). A company is included in a portfolio unless, for any day, any data regarding this company is missing. If data is missing, the company is excluded from further calculations. The sample for the first-day return calculations includes 385 (486) companies for the WSE (NC). For 30-day and 360-day returns, the sample size is 347 (395) and 332 (382) for the WSE and NC, respectively due to firms being acquired or delisted before the 30-day or 360-day period finishes.







Panel B (EW)

Figure 3. Market-adjusted short- and long-term performance of the WSE and NC IPOs from January 2005 to June 2017.

Panel A and B show the stock performance following an IPO. Returns are market-adjusted and cumulated over the relevant period using CAR procedure. The solid line depicts the performance of the WSE IPOs portfolio, while the dotted line represents performance of portfolios of the NC IPOs portfolio. Portfolios are formed using IPO-size weights winsorized at 0.5% and 99.5% (Panel A) or equal weights (Panel B). A company is included in a portfolio unless for any day, any data regarding this company is missing. If there is data missing, a company is excluded from further calculations. The sample for the first-day return calculations includes 385 (486) companies for the WSE (NC). For 30-day and 360-day returns, the WSE (NC) sample size is 347 (395) and 332 (382), respectively.





Figure 4. Cumulated percentage changes in the aggregated pension funds' AUM, in the equity part of the aggregated pension funds' AUM and in the Warsaw Stock Exchange WIG index.

Consecutive plots in the figure depict the cumulated percentage changes in the aggregated pension funds' AUM, in the equity part of the aggregated pension funds' AUM and in the Warsaw Stock Exchange WIG index. The changes are scaled relative to December 31, 2013.



Figure 5. Trend line of the mean IPO size by year in the WSE and NC from January 2005 to December 2016.

Figure 5 shows the changing mean IPO size from year on year. Panel A and B show annual mean IPO size for the WSE and NC listings, respectively. The IPO size is defined as number of shares issued multiplied by the IPO price. The IPO size are reported in million PLN. The data frame spans from January 2011 to December 2016 (Panel A) and from January 2007 to December 2016 (Panel B). Year 2017 is omitted because the sample period ends in June 30.



Figure 6. Bootstrapped pre-post differences in IPO size in the WSE and NC.

Figure 6 shows the empirical distribution of bootstrapped pre-post differences in IPO size for the WSE (Panel A) and NC (Panel B). The IPO size is defined as number of shares issued multiplied by the IPO price. The IPO size is reported in million PLN. For the purpose of bootstrapping, we sample IPOs with replacement and calculate difference between mean IPO size in the pre-treatment period and mean IPO size in the post-treatment period. We use February 3, 2014 as our cut-off date. We use a symmetric 3-year period around the cut-off in IPO sampling procedure (the short-time window). We plot the density based on 10,000 sampling iterations. The original sample differences are illustrated with the solid vertical lines.



0%

-2

-1



 0 NC IPO size pre-post difference (PLN million)

3

4

Figure 7. Placebo test for the robustness of the regulatory treatment effect on the IPO size.

Figure 7 shows the results of the placebo test of our main results. For each day (cut-off), we run the regression $Y_i = \beta_1 WSE_i + \beta_2 WSE_i \times Post_i + \gamma_t + \delta_j + \varepsilon_i$. IPO size (in million PLN) is the response variable in the regression. *WSE* is an indicator variable that takes on a value of 1 for IPO listings in the WSE market, and 0 for the WSE IPO listings. *Post* is an indicator variable that takes on a value of 1 for IPO listings post the cut-off date, and 0 for the pre-cut-off IPO listings. *WSE* × *Post* is the variable of interest which captures the difference-indifferences effect. All regressions include industry and time fixed effects. We plot the p-values of the coefficient of interest (β_2). In estimation, we use the symmetric period +/- 3 years relative to the cut-off date (short-time window; solid line) and full sample, starting on January 1, 2008 (long-time window; dotted line).





Table 1. Summary statistics for IPOs listed in the WSE and NC from January 2008 to June 2017.

Panel A contains descriptive statistics regarding IPO returns in the WSE, and Panel B contains the statistics for IPO returns in the NC. IPO size is the number of shares issued at an IPO multiplied by the IPO offer price. All returns are in decimal format; IPO size is in million PLN. Unadjusted returns are calculated using the HPR, whereas market-adjusted returns are computed using the CAR methodology. The average returns are equally weighted (EW average) or IPO-size-weighted (SW average). Our sample extends from January 2008 to June 2017. The sample size for the IPO size is 706, including 237 IPOs in the WSE and 469 IPOs in the NC. The sample diminishes in size when we analyse post-IPO short- and long-term underperformance. For reference, we report the number of observations in the last row of each panel.

			1st day		30-day		360-day		
Sample	IDO Sizo	1st day	market-	30-day	market-	360-day	market-		
statistic	IFO SIZE	return	adjusted	return	adjusted	return	adjusted		
			return		return		return		
Panel A: Summary statistics for companies listed in the WSE.									
EW Avg.	177.001	0.058	0.056	0.018	0.017	0.038	0.012		
VW Avg. ^a	-	0.055	0.056	0.049	0.048	0.121	0.042		
Std. Dev.	549.801	0.241	0.242	0.205	0.211	0.574	0.524		
Min	0.000	-0.741	-0.729	-0.716	-0.672	-0.899	-1.130		
P25	0.000	-0.023	-0.023	-0.076	-0.085	-0.293	-0.325		
P50	26.000	0.019	0.019	0.004	0.007	0.000	-0.027		
P75	96.300	0.084	0.081	0.104	0.117	0.235	0.227		
Max	3,652.757	2.420	2.448	0.783	0.912	3.451	3.250		
Obs.	237	235	235	222	222	207	207		
Panel B: Sumn	nary statistics f	or companie	s listed in the	NC.					
EW Avg.	2.609	0.400	0.401	0.431	0.430	0.126	0.072		
VW Avg. ^a	-	0.254	0.254	0.216	0.213	-0.075	-0.108		
Std. Dev.	5.721	1.505	1.506	2.976	2.979	1.845	1.828		
Min	0.000	-0.906	-0.916	-0.908	-1.016	-0.973	-1.373		
P25	0.571	0.000	0.000	-0.200	-0.192	-0.574	-0.587		
P50	1.090	0.185	0.187	0.006	0.013	-0.200	-0.253		
P75	2.580	0.518	0.512	0.343	0.351	0.169	0.167		
Max	85.138	29.000	28.992	51.000	51.010	25.331	25.132		
Obs.	469	468	467	382	381	370	369		

^a IPO size weighting with IPO size winsorized at 0.5% and 99.5%.

Table 2. IPO Sector and Year Distribution

Panel A shows the sample distribution of the WSE and NC IPOs by the GICS sector. Panel B shows the sample distribution of the WSE and NC IPOs by the IPO year. Our sample extends from January 2008 to June 2017. The sample size is 706 IPOs, including 237 IPOs in the WSE and 469 IPOs in the NC. The sample diminishes in size when we analyse post-IPO short- and long-term underperformance. For reference, we report the number of observations in the last row of each panel.

Panel A: Sample distribution by the sector									
Exchange	WSE	(no. and %)	NC	(no. and %)					
Energy	7	3.0%	2	0.4%					
Materials	18	7.6%	16	3.4%					
Industrials	45	19.0%	114	24.3%					
Consumer Discretionary	44	18.6%	110	23.5%					
Consumer Staples	17	7.2%	20	4.3%					
Health Care	18	7.6%	30	6.4%					
Financials	35	14.8%	61	13.0%					
Information Technology	24	10.1%	74	15.8%					
Telecommunication Services	1	0.4%	13	2.8%					
Utilities	8	3.4%	10	2.1%					
Real Estate	20	8.4%	19	4.1%					
TOTAL	237	100.0%	469	100.0%					

Panel B: Sample distribution by the IPO year								
Exchange	WSE	(no. and %)	NC	(no. and %)				
2008	32	13.5%	48	10.2%				
2009	13	5.5%	18	3.8%				
2010	34	14.3%	78	16.6%				
2011	34	14.3%	150	32.0%				
2012	18	7.6%	83	17.7%				
2013	23	9.7%	36	7.7%				
2014	27	11.4%	19	4.1%				
2015	29	12.2%	17	3.6%				
2016	19	8.0%	14	3.0%				
2017 ^a	8	3.4%	6	1.3%				
TOTAL	237	100.0%	469	100.0%				

^a As of June 30, 2017.

Table 3. Two-sample *t*-test of pre- and post-treatment mean returns and IPO sizes.

The table presents pre- and post-treatment mean values of the first-day, 30-day, and 360-day market-adjusted returns and IPO size for WSE and NC listings. In Panel A and B, we show the results using a short-time window implementation strategy (3-year symmetric windows; the period of February 4, 2011, till February 3, 2017) and a long-time window implementation strategy (January 1, 2008-July 30, 2017) respectively. The sample size varies depending on the length of the sample window as well as data requirements with respect to the dependent variable. For reference, we report the number of observations in the last row of each panel. Returns are market-adjusted and cumulated over the relevant period as CARs and are reported in percentages. The IPO size equals the IPO offer price multiplied by the number of shares issued and is reported in million PLN. *, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively.

	IPO Size	1st day market- adjusted return	30-day market- adjusted return	360-day market- adjusted return					
Panel A: Implementation strategy using a short-time window.									
WSE									
Pre-mean	201.248	0.023	0.023	-0.041					
Post-mean	59.220	0.035	0.022	0.060					
Difference	-142.028 **	0.012	0.000	0.101					
<i>t</i> -statistic	(2.133)	(-0.548)	(0.005)	(-1.086)					
Obs.	148	148	143	132					
NC									
Pre-mean	2.586	0.326	0.318	-0.106					
Post-mean	2.273	0.449	1.050	0.194					
Difference	-0.313	0.123	0.732	0.299					
<i>t</i> -statistic	(0.614)	(-0.916)	(-1.292)	(-1.071)					
Obs.	306	305	257	246					
Panel B: Implementation st	rategy using a lon	g-time window.							
WSE									
Pre-mean	220.200	0.067	0.019	-0.008					
Post-mean	92.224	0.033	0.013	0.060					
Difference	-127.976 **	-0.035	-0.005	0.067					
<i>t</i> -statistic	(2.222)	(1.323)	(0.191)	(-0.920)					
Obs.	237	235	222	207					
NC									
Pre-mean	2.616	0.394	0.350	0.059					
Post-mean	2.553	0.454	1.019	0.194					
Difference	-0.064	0.060	0.669	0.134					
<i>t</i> -statistic	(0.122)	(-0.579)	(-1.272)	(-0.470)					
Obs.	469	467	381	369					

Table 4. Effects of the 2014 regulatory change on the IPO underpricing, long-term underperformance and on the IPO size.

The table reports results for the regression $Y_i = \beta_1 WSE_i + \beta_2 WSE_i \times Post_i + \gamma_t + \delta_j + \varepsilon_i$. The dependent variable is either: the IPO size (in million PLN), the first-day market-adjusted return, the 30-day market-adjusted return, or the 360-day market-adjusted return (all in percentages), and our unit of observation is a single IPO. *WSE* is an indicator variable that takes on a value of 1 for IPO listings in the WSE market, and 0 for the NC IPO listings. *Post* is an indicator variable that takes on a value of 1 for post-event IPO listings, and 0 for the pre-event IPO listings, *WSE* × *Post* is the variable of interest that captures the difference-in-differences effect. All regressions include industry and year fixed effects. Year fixed effects indicate the year of the IPO. *t*-statistics are reported in parentheses. Data are for the period February 4, 2011-February 3, 2017 for Panel A, and January 1, 2008-July 30, 2017 for Panel B. The sample size varies depending on the length of the sample window as well as data requirements with respect to the dependent variable. For reference, we report the number of observations in the last row of each panel. Returns are in decimal format; IPO size is in million PLN. *, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively. We winsorize the observed IPO sizes at 0.5% and 99.5%.

	IPO Size	1st day market- adjusted return	30-day market- adjusted return	360-day market- adjusted return					
Panel A: Implementation strategy using a short-time window.									
WSE	184.960 ***	-0.213 ***	-0.154	0.077					
	(2.797)	(-4.367)	(-1.433)	(0.702)					
WSE x Post	-127.825 *	-0.174 **	-0.872	-0.219					
	(-1.895)	(-2.060)	(-1.620)	(-0.870)					
Industry FE	Yes	Yes	Yes	Yes					
Year FE	Yes	Yes	Yes	Yes					
R^2	0.136	0.040	0.032	0.060					
Obs.	454	453	400	378					
Panel B: Implemen	tation strategy using	a long-time window.							
WSE	194.130 ***	-0.317 ***	-0.271 ***	-0.135					
	(4.039)	(-5.382)	(-3.309)	(-0.930)					
WSE x Post	-106.689 **	-0.103	-0.737	-0.052					
	(-1.969)	(-1.251)	(-1.463)	(-0.202)					
Industry FE	Yes	Yes	Yes	Yes					
Year FE	Yes	Yes	Yes	Yes					
R^2	0.180	0.044	0.029	0.027					
Obs.	706	702	603	576					

Table 5. Effects of the 2014 regulatory change on the IPO underpricing, long-term underperformance and on the IPO size – alternative regression specification with additional control variables.

The table reports results for the regression $Y_i = \beta_1 WSE_i + \beta_2 WSE_i \times Post_i + \beta_3 \log (Assets)_i + \beta_4 ROA_i + \beta_5 I(VC)_i + \gamma_t + \delta_j + \varepsilon_i$. The dependent variable is either: the IPO size (in million PLN), the first-day marketadjusted return, the 30-day market-adjusted return, or the 360-day market-adjusted return (all in percentages), and our unit of observation is a single IPO. *WSE* is an indicator variable that takes on a value of 1 for IPO listings in the WSE market, and 0 for the NC IPO listings. *Post* is an indicator variable that takes on a value of 1 for postevent IPO listings, and 0 for the pre-event IPO listings, $WSE \times Post$ is the variable of interest that captures the difference-in-differences effect. log (*Assets*) are the total assets in the last fiscal year prior to the IPO. *ROA* is the return on assets in the last fiscal year prior to the IPO, using the end-of-period value of total assets in the denominator. I(VC) is an indicator of whether an IPO was Venture Capital backed. All regressions include industry and year fixed effects. Year fixed effects indicate the year of the IPO. *t*-statistics are reported in parentheses. Data are for the period February 4, 2011-February 3, 2017 for Panel A, and January 1, 2008-July 30, 2017 for Panel B. The sample size varies depending on the length of the sample window as well as data requirements for the dependent variable. For reference, we report the number of observations in the last row of each panel. Returns are in decimal format; IPO size is in million PLN. *, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively. We winsorize the observed IPO sizes at 0.5% and 99.5%.

	IPO Size	1st day market- adjusted return	30-day market- adjusted return	360-day market- adjusted return					
Panel A: Implementation strategy using a short-time window.									
WSE	35.260	-0.068 *	0.052	0.089					
	(1.405)	(-1.944)	(0.392)	(0.653)					
WSE x Post	-96.477 *	-0.238 ***	-1.009 *	-0.235					
	(-1.808)	(-3.876)	(-1.724)	(-0.895)					
Controls	Yes	Yes	Yes	Yes					
Industry FE	Yes	Yes	Yes	Yes					
Year FE	Yes	Yes	Yes	Yes					
R2	0.195	0.198	0.099	0.085					
Obs.	415	414	371	351					
Panel B: Implementatio	n strategy using a l	ong-time window.							
WSE	-39.346	-0.171 ***	-0.082	0.040					
	(-1.214)	(-3.328)	(-0.911)	(0.384)					
WSE x Post	-61.227	-0.172 **	-0.881	-0.208					
	(-1.378)	(-2.487)	(-1.643)	(-0.833)					
Controls	Yes	Yes	Yes	Yes					
Industry FE	Yes	Yes	Yes	Yes					
Year FE	Yes	Yes	Yes	Yes					
R2	0.302	0.117	0.087	0.062					
Obs.	643	639	556	531					

Table 6. Issuer effects of the 2014 regulatory change on the IPO underpricing, long-term underperformance and on the IPO size across different IPO size sample partitions.

The table reports difference-in-differences estimates for the regression $Y_i = \beta_1 WSE_i + \beta_2 WSE_i \times Post_i + \gamma_t + \gamma_t$ $\delta_i + \varepsilon_i$. We estimate the regression for the IPO size partitions. To construct the size partitions, we sort the sample based on IPO size each year. Prior to sorting, we exclude issuers with IPO size unavailable or reported as 0 by the exchange. Small issuers are issuers in the smallest size quartile (bottom 25% issuers). Big issuers are issuers in the largest size quartile (top 20% issuers). We also report intermediate portfolios (labelled portfolio 2 and 3) corresponding to the second and third quartile of IPO size. The dependent variable is either: the IPO size (in million PLN), the first-day market-adjusted return, the 30-day market-adjusted return, or the 360-day marketadjusted return (all in percentages), and our unit of observation is a single IPO. Log IPO size is the natural logarithm of IPO size. In the regression, WSE is an indicator variable that takes on a value of 1 for IPO listings in the WSE market, and 0 for the NC IPO listings. Post is an indicator variable that takes on a value of 1 for postevent IPO listings, and 0 for the pre-event IPO listings, $WSE \times Post$ is the variable of interest that captures the difference-in-differences effect. All regressions include industry and year fixed effects. Year fixed effects indicate the year of the IPO. We only report difference-in-differences estimates, which are the coefficients on WSE × Post. t-statistics are reported in parentheses. Data are for the period February 4, 2011-February 3, 2017 for Panel A, and January 1, 2008-July 30, 2017 for Panel B. The sample size varies depending on the length of the sample window as well as data requirements for the dependent variable. Returns are in decimal format; IPO size is in million PLN. *, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively. We winsorize observed IPO sizes at 0.5% and 99.5%.

			1st day	30-day	360-day					
	IDO Sizo	Log IPO	market-	market-	market-					
	IFO SIZE	Size	adjusted	adjusted	adjusted					
			return	return	return					
Panel A: Implementation strategy using a short-time window.										
Small	-6.616	-0.483	-0.254 *	-0.203	-0.737					
	(-1.602)	(-1.361)	(-1.865)	(-0.526)	(-1.196)					
2	-6.390	-0.068	-0.173	-0.063	0.264					
	(-1.102)	(-0.506)	(-1.122)	(-0.196)	(1.450)					
3	-24.493 **	-0.596 **	-0.405	-0.416	-0.704					
	(-2.250)	(-2.450)	(-0.956)	(-0.533)	(-1.113)					
Big	-669.324 **	-0.870 **	-0.300 **	-0.694 *	0.086					
	(-2.354)	(-2.285)	(-2.287)	(-1.664)	(0.276)					
Panel B: Implementation st	rategy using a l	ong-time wind	low.							
Small	1.639	0.267	-0.214	-0.063	-0.388					
	(0.482)	(0.663)	(-1.320)	(-0.174)	(-0.560)					
2	1.542	0.248	-0.178	0.155	0.544					
	(0.300)	(1.451)	(-0.843)	(0.525)	(1.074)					
3	31.009	-0.082	0.010	-0.134	-0.220					
	(1.199)	(-0.317)	(0.034)	(-0.246)	(-0.377)					
Big	-634.297 **	-0.844 **	-0.241 **	-0.631 *	0.178					
	(-2.343)	(-2.449)	(-2.114)	(-1.664)	(0.598)					

Appendix A

In this appendix we describe the IPO datasets in greater detail in order to facilitate the replication and extension of our study.

IPO Data

We obtain data on IPOs in the Polish stock market from the following providers: GPW (*gpw.pl* and *newconnect.pl*), Stooq (*stooq.com*), Bossa (*bossa.pl*) and Bloomberg.

GPW data available at *gpw.pl*, the official website of the Warsaw Stock Exchange ('WSE'), is the publicly available source of information about securities listed in the WSE (*gpw.pl*). Similarly, the GPW also provides the data about securities listed in the New Connect ('NC') equity market. (*newconnect.pl*). GPW data includes the complete list of IPOs for both markets as well as the summary of the most relevant IPO data. The IPO data available directly from GPW are IPO date, IPO size, and IPO offer price. In rare cases (<1% of the sample), where the GPW provides two offer prices for the same IPO corresponding to different share series, we verify IPO prospectuses individually. We only use GPW's prices to calculate first-day returns but not to calculate IPO size, which is provided by GPW separately.

The IPO data from GPW also includes the transitions from the NC to the WSE market. In cases where the NC company opts for a direct listing in the WSE without new issuance, the GPW assigns it the IPO size of 0, and we retain this treatment as appropriate.

Stock Prices

We retrieve the adjusted daily stock prices from the Stooq database. Adjustments account for stock splits, dividends, and other distributions, rights offerings, and denominations. This data is publicly available at *stooq.com*. The same data was previously used in the studies of the Polish IPO market (e.g. Jewartowski & Lizińska, 2012). We note that in the Stooq database, IPO price definition is not consistent throughout the sample: sometimes first observation is not the IPO price, but a first-day close price. We use IPO offer prices retrieved from the GPW database to verify the starting data point in the Stooq database. We retrieve non-adjusted (raw) daily stock prices from the Bossa database. This is publicly available data provided by Brokerage Unit of Bank Ochrony Środowiska S.A. Bossa provides time-series data in the textual files, separately for each security. The data is publicly available at *bossa.pl*.

Bloomberg is used to obtaining any missing time series data, particularly the daily adjusted and non-adjusted time series of the stock prices for delisted, acquired, and liquidated companies.

In our dataset, there are 178 companies with a market status other than 'currently active', including 58 delisted companies in the WSE and 120 delisted companies in the NC. We do not rely on Bloomberg as our main data source because Bloomberg's coverage of Polish IPOs starts with a few days delay relative to the actual IPO date and is not backfilled. As a result, early data is missing. Moreover, not all IPOs (especially from the NC) are included in Bloomberg's data.

Financial Statements Data

We obtain financial statements data from Notoria Serwis S.A. ("Notoria"). The access to the Notoria database is not public, although the financial statements are available from companies' websites. Notoria Financials offers high-quality financial statements data in the standardized format. The database also includes data from the reporting periods prior to the IPO (as reported by the company).

Moreover, we use Notoria to access header information such as the company's name, trading ticker, or the International Security Identifier Number (ISIN). We manually merge IPO data with financial statements data using the company name. Historical names for the companies that experienced name change are stored in the underlying Notoria files, enabling us to ensure the accuracy of matches. We then use company trading ticker and ISIN to merge the remaining datasets.

Other Data

We use the Dealogic database to access the information about IPO's Venture-Capital backing status.

We use a one-month Warsaw Interbank Offer Rate (WIBOR) as a proxy for the risk-free rate. We obtain WIBOR quotes from *stooq.com*. The rate is available with ticker PLOPLN1M.

We use WIG and NCIndex indices to track the performance of stocks in the main and the NC markets. We obtain WIG and NCIndex quotes from *stooq.com*. We use index levels to calculate returns, defined as a percentage change in the index value. The quotes are available with tickers WIG and NCIndex, respectively.

We use sector betas provided by Aswath Damodaran. The data is publicly available via Damodaran's NYU website. We use the July 1, 2017 update of the database, which corresponds to the end of our sample period. We use a leveraged beta database for Europe, which includes Polish stocks. We link companies to the industries using the industry details file, available in *Data: Breakdown* section of Damodaran's website.

Links

- WSE IPOs list <u>https://www.gpw.pl/debiuty</u>
- NC IPOs list <u>https://newconnect.pl/debiuty-spolek</u>
- Adjusted prices <u>https://stooq.com/</u>
- Unadjusted prices https://info.bossa.pl/notowania/metastock/
- Sector betas <u>http://pages.stern.nyu.edu/~adamodar/</u>

Appendix B

In this Appendix we provide times-series and cross-section of first-day IPO returns in the Polish stock market.

Time-Series of First-Day Returns

Table A2 shows the summary statistics for market-adjusted returns for companies listed on the WSE (Panel A) and NC (Panel B) for each year from January 2005 to June 2017. We note that the NC first-day returns are higher than the corresponding WSE returns, except in the year 2007, when the NC market launched and when average IPO first-day returns were plummeting. Loughran and Ritter (2004) suggest that general market conditions explain first-day returns. They argue that IPO underpricing effect is sample- and period-specific, and report that in the U.S., IPOs were underpriced by an average of 18% in the 1990s, 65% during the Internet bubble (1999–2000), while only 12% in the following years 2000–2003.

[Insert Table A2. here]

Cross-Section of First-Day Returns

Table A3 shows the summary statistics for the market-adjusted first-day returns for companies listed in the WSE (Panel A) and NC (Panel B), broken down into 11 GICS sectors.

[Insert Table A3. here]

Financials, Energy, and Healthcare sectors experience the highest degree of IPO underpricing. The average first-day market-adjusted return equals 17.67% for companies in the financial sector, 17.17% for those in the energy sector and 14.79% for those in the healthcare. When size weights are used, the returns are 13.37%, 20% and 18.23%, respectively. The high first-day returns for the energy and healthcare sector might be linked to the demand argument. According to the theory of market rotation proposed by Bodie et al. (2014), investors can see when an economy is at its peak and anticipate contraction or recession (which was likely, given the level of uncertainty during the global financial crisis). Thus, they tend to invest in noncyclical industries such as energy or healthcare.

The NC differs significantly from the WSE in terms of the cross-section of IPO returns. First, most of the estimates are statistically significant. Second, information technology (IT), materials and telecommunication are the most prominent sectors yielding the highest first-day returns in the NC market. This is not surprising given the NC profile. IT stocks earn the highest equally weighted market-adjusted first-day return, 79.73% (as compared to the IPO-size-

weighted market-adjusted return, 45.93%). We relate this outperformance to the divergence of opinions. Many promising small firms take advantage of the low barrier to entry into the NC market and the fact that investing in technological start-ups is in fashion. However, there is little pre-IPO information on the performance of start-up companies, which causes high divergence of opinions and eventually results in higher underpricing, as stated by Miller (1977). Third, real estate and industrial companies are the least underpriced during the IPO process in the NC market.

Damodaran database	•												
Sample statistic	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017 ^a
Panel A: Summary statistics for companies listed in the WSE.													
No. companies	33	38	79	32	13	34	34	18	23	27	29	19	6
EW average	0.0796	0.3476	0.1498	0.1253	0.1730	0.0683	0.0057	0.0354	0.0414	0.0215	0.0319	0.0675	-0.0107
SW average ^b	0.1345	0.1681	0.0985	0.0153	0.1441	0.0891	0.0082	0.0040	0.0598	0.0278	0.0481	0.0587	0.0659
Median	0.0385	0.1083	0.0653	0.0103	0.1384	0.0231	0.0130	0.0066	0.0022	0.0198	0.0082	0.0439	0.0111
Minimum	-0.1007	-0.0210	-0.1773	-0.7286	-0.0410	-0.1497	-0.2913	-0.6237	-0.3107	-0.1181	-0.1286	-0.0522	-0.2283
Maximum	0.6123	4.7890	1.6211	2.4478	0.5283	1.0703	0.2767	0.6665	0.2646	0.1251	0.7016	0.4365	0.1071
1 st quartile	0.0091	0.0266	-0.0203	-0.0665	0.0527	-0.0117	-0.0069	-0.0105	-0.0335	0.0018	-0.0504	-0.0133	-0.0259
3 rd quartile	0.1480	0.2878	0.2325	0.1020	0.2466	0.1355	0.0540	0.0647	0.1723	0.0505	0.0591	0.0946	0.0575
Std dev	0.1435	0.8068	0.2764	0.5286	0.1629	0.2013	0.1100	0.2444	0.1367	0.0518	0.1475	0.1209	0.1178
t-statistic	3.1866	2.6556	4.8175	1.3410	3.8307	1.9801	0.3015	0.6152	1.4522	2.1566	1.1662	2.4321	-0.2228
<i>p</i> -value	0.0032	0.0116	0.0000	0.1897	0.0024	0.0561	0.7649	0.5466	0.1605	0.0405	0.2534	0.0257	0.8325
Panel B: Summary	y statistics for	r companie:	s listed in tl	ne NC.									
No. companies			18	48	18	78	150	82	36	19	17	14	5
EW average			-0.2798	0.6159	0.9543	0.3295	0.2190	0.5821	0.2828	0.3938	0.4204	0.4559	0.5928
SW average ^b			-0.1031	0.2268	0.3028	0.2933	0.1199	0.4329	0.3506	0.5258	0.6761	0.3985	0.2294
Median			-0.2872	0.0711	0.3033	0.2499	0.1332	0.2018	0.2101	0.2959	0.1125	0.4051	0.2988
Minimum			-0.9295	-0.3209	-0.6045	-0.9156	-0.6894	-0.5686	-0.3539	-0.2008	-0.3982	-0.4310	0.0957
Maximum			0.2547	6.3110	9.0065	2.0977	1.2454	28.9923	1.2335	1.2909	1.4389	1.2515	1.2338
1 st quartile			-0.5907	-0.0634	0.0360	0.0041	-0.0528	0.0029	0.0150	0.0277	0.0227	0.2370	0.1488
3 rd quartile			0.0200	0.5985	1.1329	0.5725	0.4971	0.4936	0.5142	0.5393	0.8606	0.6790	1.1871
Std dev			0.3670	1.3426	2.1238	0.4466	0.3625	3.1948	0.3718	0.4689	0.5621	0.4172	0.5689
t-statistic			-3.2339	3.1784	1.9063	6.5158	7.3975	1.6499	4.5642	3.6607	3.0839	4.0888	2.3300
<i>p</i> -value			0.0049	0.0026	0.0737	0.0000	0.0000	0.1028	0.0001	0.0018	0.0071	0.0013	0.0803

Table B1. Summary statistics for market-adjusted first-day returns in the WSE and NC for each year from January 2005 to June 2017.

Returns are market-adjusted against the expected return determined by CAPM for each observation. We use Warsaw Interbank Offer Rate (WIBOR) rates as a proxy for the risk-free rate. We use main index, the Warsaw Stock Exchange Index (WIG), as a proxy for market portfolio. Our estimated betas are based on levered industry betas from Damodaran database.

^a As of June 30, 2017.

^b IPO size weighting with IPO size winsorized at 0.5% and 99.5%.

Table B2. Summary statistics for market-adjusted first-day returns in the WSE and NC, sorted by sector (aggregated industry).

Returns are market-adjusted against the expected return determined by CAPM for each observation. We use Warsaw Interbank Offer Rate (WIBOR) rates as a proxy for the risk-free rate. We use the main index, the Warsaw Stock Exchange Index (WIG), as a proxy for the market portfolio. Our estimated betas are based on levered industry betas from Damodaran database. For the purpose of market adjustment (beta adjustment), we assign each company to one of 94 sectors. For convenience, here we aggregate the industries into 11 sectors using the GICS.

Sample statistic	Consumer discret.	Consumer staples	Energy	Financials	Health care	Industrials	IT	Materials	Real estate	Telecom.	Utilities
Panel A: Summary statistics for companies listed in the WSE.											
No. companies	80	37	11	43	24	80	41	25	28	5	11
EW average	0.0947	0.0413	0.1717	0.1767	0.1479	0.0937	0.1225	0.0063	0.1809	0.1174	0.0271
SW average ^a	0.0860	0.0569	0.2000	0.1337	0.1823	0.1195	0.1253	0.0048	0.0436	0.1410	0.0424
Median	0.0437	0.0439	0.0290	0.0314	0.0642	0.0309	0.0272	0.0050	0.0489	0.1430	0.0140
Minimum	-0.1773	-0.7286	-0.2913	-0.1497	-0.0634	-0.2253	-0.1286	-0.2847	-0.6237	-0.0573	-0.0693
Maximum	0.9892	0.3768	1.6211	4.7890	1.3252	1.4650	0.7096	0.2767	2.4478	0.3430	0.1384
1 st quartile	-0.0070	-0.0124	0.0016	-0.0087	0.0105	-0.0053	-0.0460	-0.0311	-0.0060	-0.0056	-0.0128
3 rd quartile	0.1148	0.1174	0.2117	0.1058	0.1970	0.1577	0.1607	0.0397	0.1559	0.1637	0.0659
Std dev	0.2072	0.1644	0.5114	0.7400	0.2745	0.2292	0.2473	0.1076	0.5327	0.1576	0.0636
<i>t</i> -statistic	4.0861	1.5271	1.1132	1.5659	2.6391	3.6562	3.1726	0.2932	1.7965	1.6652	1.4123
<i>p</i> -value	0.0001	0.1355	0.2916	0.1249	0.0147	0.0005	0.0029	0.7719	0.0836	0.1712	0.1882
Panel B: Summary sta	tistics for con	npanies listed	in the NC.								
No. companies	112	21	2	62	30	118	78	16	22	14	10
EW average	0.3576	0.2337	0.1545	0.3421	0.4192	0.1941	0.7973	0.4222	0.1742	0.3659	0.2296
SW average ^a	0.2612	0.1958	0.2085	0.1517	0.1699	0.1617	0.4593	0.4250	0.0698	0.4075	0.2571
Median	0.1754	0.2262	0.1545	0.1522	0.0729	0.0871	0.3540	0.3782	0.1668	0.3210	0.1610
Minimum	-0.6638	-0.8159	0.0682	-0.6216	-0.4310	-0.9156	-0.9295	-0.6894	-0.3680	-0.3332	-0.0973
Maximum	3.4600	0.8293	0.2408	6.3110	9.0065	1.7978	28.9923	1.2909	1.0036	1.2483	0.7456
1 st quartile	-0.0012	0.0592	0.1113	0.0007	-0.0773	-0.0523	0.0140	0.2023	-0.0068	0.0981	-0.0538
3 rd quartile	0.5873	0.5060	0.1976	0.4965	0.2960	0.4183	0.6899	0.7134	0.3177	0.5277	0.4784
Std dev	0.6180	0.3930	0.1220	0.8567	1.6563	0.4260	3.3177	0.4833	0.3273	0.4382	0.3127
t-statistic	6.1228	2.7249	1.7905	3.1442	1.3863	4.9485	2.1224	3.4941	2.4962	3.1240	2.3222
<i>p</i> -value	0.0000	0.0130	0.3243	0.0026	0.1762	0.0000	0.0370	0.0033	0.0209	0.0081	0.0453

^a IPO size weighting with IPO size winsorized at 0.5% and 99.5%.

Appendix C

In this Appendix, we provide additional analyses and robustness checks, as referenced throughout the paper.

Figure C1. Sample composition by industry and year.

Figure C1 shows the composition of the sample IPOs by industry and year. Our sample extends from January 2008 to June 2017. The sample size is 706 IPOs, including 237 IPOs in the WSE and 469 IPOs in the NC.



Figure C2. Pre-post differences in IPO size in the WSE and NC from the permutation test.

Figure C2 shows the distribution of pre-post differences in IPO size for the WSE (Panel A) and NC (Panel B) based on the permutation test. The IPO size is defined as the number of shares issued multiplied by the IPO price. The IPO size is reported in million PLN. For the purpose of permutations, we sample IPOs without replacement and calculate the difference between mean IPO size in the post-treatment period and mean IPO size in the pre-treatment period. We use February 3, 2014, as our cut-off date. We use a symmetric 3-year period around the cut-off in the IPO sampling procedure (the short-time window). We plot the density based on 10,000 sampling iterations. The original sample differences are illustrated with the solid vertical lines.



Panel A (WSE)

Table C1. Expanded Summary statistics for IPOs listed in the WSE and NC from January 2008 to June2017.

Panel A contains descriptive statistics regarding IPO returns in the WSE, and Panel B contains the statistics for IPO returns in the NC. The sample for the first-day return calculations includes 385 (486) companies for the WSE (NC). For 30-day and 360-day returns, the WSE (NC) sample size is 347 (395) and 332 (382), respectively. Unadjusted returns are calculated using the HPR, whereas market-adjusted returns are computed using the CAR methodology. The average returns are equally weighted (EW average) or IPO-size-weighted (SW average). IPO size is the number of shares issued at an IPO multiplied by the IPO offer price. All returns are in decimal format; IPO size is in million PLN.

			1st day		30-day		360-day		
Sample	IDO Sizo	1st day	market-	30-day	market-	360-day	market-		
statistic	IFO SIZE	return	adjusted	return	adjusted	return	adjusted		
			return		return		return		
Panel A: Summary statistics for companies listed in the WSE.									
EW Avg.	215.017	0.058	0.056	0.018	0.017	0.038	0.012		
VW Avg.		0.055	0.056	0.049	0.048	0.121	0.042		
Std. Dev.	827.895	0.241	0.242	0.205	0.211	0.574	0.524		
Min	0.000	-0.741	-0.729	-0.716	-0.672	-0.899	-1.130		
P1	0.000	-0.287	-0.304	-0.496	-0.517	-0.815	-0.912		
P5	0.000	-0.137	-0.132	-0.294	-0.293	-0.697	-0.685		
P10	0.000	-0.070	-0.074	-0.200	-0.212	-0.582	-0.561		
P25	0.000	-0.023	-0.023	-0.076	-0.085	-0.293	-0.325		
P50	26.000	0.019	0.019	0.004	0.007	0.000	-0.027		
P75	96.300	0.084	0.081	0.104	0.117	0.235	0.227		
P90	294.402	0.208	0.211	0.246	0.245	0.606	0.563		
P95	693.232	0.282	0.268	0.333	0.349	0.926	0.781		
P99	4,954.033	0.947	0.945	0.681	0.735	1.871	1.623		
Max	8,068.543	2.420	2.448	0.783	0.912	3.451	3.250		
Ν	237	235	235	222	222	207	207		
Panel B: Sum	mary statistics f	or companie	s listed in the	NC.					
EW Avg.	2.609	0.400	0.401	0.431	0.430	0.126	0.072		
VW Avg.	0.000	0.254	0.254	0.216	0.213	-0.075	-0.108		
Std. Dev.	5.721	1.505	1.506	2.976	2.979	1.845	1.828		
Min	0.000	-0.906	-0.916	-0.908	-1.016	-0.973	-1.373		
P1	0.000	-0.463	-0.455	-0.773	-0.749	-0.934	-1.159		
P5	0.102	-0.269	-0.269	-0.568	-0.582	-0.858	-0.987		
P10	0.260	-0.157	-0.153	-0.423	-0.406	-0.782	-0.809		
P25	0.571	0.000	0.000	-0.200	-0.192	-0.574	-0.587		
P50	1.090	0.185	0.187	0.006	0.013	-0.200	-0.253		
P75	2.580	0.518	0.512	0.343	0.351	0.169	0.167		
P90	5.567	0.862	0.867	1.162	1.163	1.021	0.867		
P95	9.558	1.197	1.195	2.033	2.076	1.752	1.663		
P99	19.778	3.237	3.234	5.346	5.282	7.991	7.750		
Max	85.138	29.000	28.992	51.000	51.010	25.331	25.132		
Ν	469	468	467	382	381	370	369		

^a IPO size weighting with IPO size winsorized at 0.5% and 99.5%.

Table C2. Two-sample nonparametric tests of pre-post treatment differences in the mean IPO size.

The table presents the results of the nonparametric tests of pre-post treatment differences in the mean IPO size for the WSE and NC listings. In Panel A and B, we show the results using a short-time window implementation strategy (3-year symmetric windows; the period of February 4, 2011, till February 3, 2017) and a long-time window implementation strategy (January 1, 2008-July 30, 2017) respectively. We use February 3, 2014, as our cut-off date. To estimate *p*-values in the bootstrapping (permutations) test, we sample IPOs with (without) replacement and calculate the difference between mean IPO size in the post-treatment period and mean IPO size in the pre-treatment period. We estimate *p*-values as the frequency of differences in the sampled data that fall below the observed sample difference. Therefore, low *p*-values imply a high statistical significance of the observed differences relative to the random benchmark. We use a symmetric 3-year period around the cut-off in the IPO sampling procedure (the short-time window). We base our estimates on the 10,000 sampling iterations. Returns are market-adjusted and cumulated over the relevant period as CARs and are reported in percentages. The IPO size equals the IPO offer price multiplied by the number of shares issued and is reported in million PLN. The *t*-statistics corresponding to the estimated p-values are reported in parentheses.

	IPO Size	1st day market- adjusted return	30-day market- adjusted return	360-day market- adjusted return						
Panel A: Implementation s	Panel A: Implementation strategy using a short-time window.									
WSE										
Bootstrapping	0.019	0.706	0.499	0.860						
	(2.073)	(-0.541)	(0.002)	(-1.079)						
Permutations	0.000	0.782	0.494	0.930						
	(3.433)	(-0.778)	(0.014)	(-1.474)						
NC										
Bootstrapping	0.369	0.841	0.873	0.897						
	(0.335)	(-0.999)	(-1.139)	(-1.265)						
Permutations	0.347	0.838	0.859	0.904						
	(0.393)	(-0.984)	(-1.077)	(-1.304)						
Panel B: Implementation s	trategy using a lo	ong-time window.								
WSE										
Bootstrapping	0.041	0.144	0.427	0.796						
	(1.742)	(1.064)	(0.185)	(-0.826)						
Permutations	0.006	0.072	0.412	0.861						
	(2.531)	(1.462)	(0.222)	(-1.083)						
NC										
Bootstrapping	0.548	0.756	0.888	0.741						
	(-0.121)	(-0.695)	(-1.215)	(-0.647)						
Permutations	0.566	0.790	0.880	0.751						
	(-0.165)	(-0.806)	(-1.173)	(-0.678)						

Table C3. Two-sample *t*-test of pre- and post-treatment mean unadjusted returns.

The table presents pre- and post-treatment mean values of the first-day, 30-day, and 360-day unadjusted (raw) returns for WSE and NC listings. In Panel A and B, we show the results using a short-time window implementation strategy (3-year symmetric windows; the period of February 4, 2011, till February 3, 2017) and a long-time window implementation strategy (January 1, 2008-July 30, 2017) respectively. The sample size varies depending on the length of the sample window as well as data requirements with respect to the dependent variable. For reference, we report the number of observations in the last row of each panel. Returns are cumulated over the relevant period as CARs and are reported in percentages.

	1st day return	30-day return	360-day return
Panel A: Implementation	strategy using a short-time	window.	
WSE			
Pre-mean	0.026	0.033	0.034
Post-mean	0.038	0.018	0.083
Difference	0.012	-0.015	0.049
<i>t</i> -statistic	(-0.542)	(0.467)	(-0.511)
Obs.	148	143	132
NC			
Pre-mean	0.325	0.314	-0.012
Post-mean	0.448	1.057	0.199
Difference	0.123	0.743	0.212
<i>t</i> -statistic	(-0.916)	(-1.312)	(-0.745)
Obs.	306	258	247
Panel B: Implementation	strategy using a long-time	window.	
WSE			
Pre-mean	0.069	0.021	0.020
Post-mean	0.035	0.010	0.083
Difference	-0.035	-0.011	0.063
<i>t</i> -statistic	(1.324)	(0.400)	(-0.803)
Obs.	235	222	207
NC			
Pre-mean	0.393	0.349	0.119
Post-mean	0.452	1.027	0.199
Difference	0.059	0.678	0.081
<i>t</i> -statistic	(-0.576)	(-1.290)	(-0.279)
Obs.	468	382	370

Table C4. Two-sample *t*-test of pre- and post-treatment mean returns and IPO sizes – alternative sample excluding 2008-2009 Financial Crisis.

The table presents pre- and post-treatment mean values of the IPO size, the first-day, 30-day, and 360-day marketadjusted returns for the WSE and NC listings. In Panel A and B, we show the results using a short-time window implementation strategy (3-year symmetric windows; the period of February 4, 2011, till February 3, 2017) and a long-time window implementation strategy (January 1, 2010-July 30, 2017) respectively. The sample excludes IPOs from 2008 and 2009 corresponding to the global financial crisis. The sample size varies depending on the length of the sample window as well as data requirements with respect to the dependent variable. For reference, we report the number of observations in the last row of each panel. Returns are market-adjusted and cumulated over the relevant period as CARs and are reported in percentages. The IPO size equals the IPO offer price multiplied by the number of shares issued and is reported in million PLN.

	IPO Size	1st day market- adjusted return	30-day market- adjusted return	360-day market- adjusted return			
Panel A: Implementation strategy using a short-time window.							
WSE							
Pre-mean	201.248	0.023	0.023	-0.041			
Post-mean	59.220	0.035	0.022	0.060			
Difference	-142.028 **	0.012	0.000	0.101			
<i>t</i> -statistic	(2.133)	(-0.548)	(0.005)	(-1.086)			
Obs.	148	148	143	132			
NC							
Pre-mean	2.586	0.326	0.318	-0.106			
Post-mean	2.273	0.449	1.050	0.194			
Difference	-0.313	0.123	0.732	0.299			
<i>t</i> -statistic	(0.614)	(-0.916)	(-1.292)	(-1.071)			
Obs.	306	305	257	246			
Panel B: Implementation strategy using a long-time window.							
WSE							
Pre-mean	233.175	0.038	0.017	-0.067			
Post-mean	92.224	0.033	0.013	0.060			
Difference	-140.951 **	-0.006	-0.004	0.127			
<i>t</i> -statistic	(2.080)	(0.277)	(0.140)	(-1.645)			
Obs.	192	190	181	166			
NC							
Pre-mean	2.679	0.335	0.336	0.013			
Post-mean	2.553	0.454	1.019	0.194			
Difference	-0.126	0.119	0.683	0.180			
<i>t</i> -statistic	(0.230)	(-1.098)	(-1.276)	(-0.622)			
Obs.	403	401	326	314			

Table C5. Effects of the 2014 regulatory change on the IPO size - winsorization choices.

The table reports results for the regression $Y_i = \beta_1 WSE_i + \beta_2 WSE_i \times Post_i + \gamma_t + \delta_j + \varepsilon_i$. The dependent variable is the IPO size (in million PLN), and our unit of observation is a single IPO. Different columns correspond to different winsorization choices. We winsorize IPO size at 0% (no winsorization) at 0.5% (0.5% and 99.5%) and at 1% (1% and 99%). *WSE* is an indicator variable that takes on a value of 1 for IPO listings in the WSE market, and 0 for the NC IPO listings. *Post* is an indicator variable that takes on a value of 1 for post-event IPO listings, and 0 for the pre-event IPO listings, *WSE* × *Post* is the variable of interest that captures the difference-in-differences effect. All regressions include industry and year fixed effects. Year fixed effects indicate the year of the IPO. *t*-statistics are reported in parentheses. Data are for the period February 4, 2011-February 3, 2017 for Panel A, and January 1, 2008-July 30, 2017 for Panel B. The sample size varies depending on the length of the sample window as well as data requirements with respect to the dependent variable. For reference, we report the number of observations in the last row of each panel. IPO size is in million PLN. *, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively.

	IPO Size (winsorized at 0%)	IPO Size (winsorized at 0.5%)	IPO Size (winsorized at 1%)				
Panel A: Implementation strategy using a short-time window.							
WSE	209.168 **	184.950 ***	147.984 ***				
	(2.412)	(2.797)	(3.457)				
WSE x Post	-151.769 *	-127.814 *	-90.945 **				
	(-1.740)	(-1.895)	(-2.029)				
Industry FE	Yes	Yes	Yes				
Year FE	Yes	Yes	Yes				
R2	0.114	0.136	0.169				
Obs.	454	454	454				
Panel B: Implementation strategy us	sing a long-time windo	w.					
WSE	247.566 ***	194.113 ***	144.526 ***				
	(3.271)	(4.040)	(4.910)				
WSE x Post	-163.203 **	-106.672 **	-56.522				
	(-2.007)	(-1.968)	(-1.468)				
Industry FE	Yes	Yes	Yes				
Year FE	Yes	Yes	Yes				
R2	0.129	0.180	0.210				
Obs.	706	706	706				

Table C6. Effects of the 2014 regulatory change on the IPO underpricing, long-term underperformance and on the IPO size – alternative regression specification with Post variable.

The table reports results for the regression $Y_i = \beta_1 WSE_i + \beta_2 Post_i + \beta_3 WSE_i \times Post_i + \delta_j + \varepsilon_i$. The dependent variable is either: the IPO size (in million PLN), a first-day market-adjusted return, a 30-day market-adjusted return, or a 360-day market-adjusted return (all in percentages), and our unit of observation is a single IPO. *WSE* is an indicator variable that takes on a value of 1 for IPO listings in the WSE market, and 0 for the NC IPO listings. *Post* is an indicator variable that takes on a value of 1 for post-event IPO listings, and 0 for the pre-event IPO listings, *WSE* × *Post* is the variable of interest that captures the difference-in-differences effect. All regressions include industry fixed effects. *t*-statistics are reported in parentheses. Data are for the period February 4, 2011-February 3, 2017 for Panel A, and January 1, 2008-July 30, 2017 for Panel B. The sample size varies depending on the length of the sample window as well as data requirements with respect to the dependent variable. For reference, we report the number of observations in the last row of each panel. Returns are in decimal format; IPO size is in million PLN. *, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively. We winsorize the observed IPO sizes at 0.5% and 99.5%.

	IPO Size	1st day market- adjusted return	30-day market- adjusted return	360-day market- adjusted return			
Panel A: Implementation strategy using a short-time window.							
WSE	187.370 ***	-0.216 ***	-0.163	0.098			
	(2.904)	(-3.364)	(-1.209)	(0.849)			
Post	1.341	0.080	0.678	0.256			
	(0.139)	(0.513)	(1.156)	(0.931)			
WSE x Post	-133.946 *	-0.189 *	-0.878	-0.216			
	(-1.961)	(-1.697)	(-1.560)	(-0.757)			
Industry FE	Yes	Yes	Yes	Yes			
Year FE	No	No	No	No			
R2	0.134	0.032 0.024		0.040			
Obs.	454	453 400 3		378			
Panel B: Implementation	strategy using a lo	ong-time window.					
WSE	194.675 ***	-0.280 ***	-0.258 **	-0.066			
	(4.082)	(-4.401)	(-2.450)	(-0.499)			
Post	18.514	0.041	0.637	0.140			
	(1.631)	(0.357)	(1.197)	(0.501)			
WSE x Post	-112.248 **	-0.151	-0.747	-0.094			
	(-1.978)	(-1.551)	(-1.430)	(-0.335)			
Industry FE	Yes	Yes	Yes	Yes			
Year FE	No	No	No	No			
R2	0.171	0.032	0.020	0.011			
Obs.	706	702	603	576			

Table C7. Effects of the 2014 regulatory change on the IPO underpricing, long-term underperformance and on the IPO size – alternative sample excluding Consumer Discretionary sector.

The table reports results for the regression $Y_i = \beta_1 WSE_i + \beta_2 WSE_i \times Post_i + \gamma_t + \delta_j + \varepsilon_i$. The dependent variable is either: the IPO size (in million PLN), the first-day market-adjusted return, the 30-day market-adjusted return, or the 360-day market-adjusted return (all in percentages), and our unit of observation is a single IPO. *WSE* is an indicator variable that takes on a value of 1 for IPO listings in the WSE market, and 0 for the NC IPO listings. *Post* is an indicator variable that takes on a value of 1 for post-event IPO listings, and 0 for the pre-event IPO listings, *WSE* × *Post* is the variable of interest that captures the difference-in-differences effect. All regressions include industry and year fixed effects. Year fixed effects indicate the year of the IPO. *t*-statistics are reported in parentheses. Data are for the period February 4, 2011-February 3, 2017 for Panel A, and January 1, 2008-July 30, 2017 for Panel B. The sample excludes IPOs from the Consumer Discretionary sector due to the observed decline in the number of IPOs in that sector and its large contribution to the full sample. The sample size varies depending on the length of the sample window as well as data requirements with respect to the dependent variable. For reference, we report the number of observations in the last row of each panel. Returns are in decimal format; IPO size is in million PLN. *, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively. We winsorize the observed IPO sizes at 0.5% and 99.5%.

	IPO Size	1st day market- adjusted return	30-day market- adjusted return	360-day market- adjusted return		
Panel A: Implementation strategy using a short-time window.						
WSE	217.085 ***	-0.212 ***	-0.143	0.144		
	(2.694)	(-3.697)	(-1.136)	(1.266)		
WSE x Post	-157.819 *	-0.207 **	-0.362 *	-0.304		
	(-1.909)	(-2.194)	(-1.756)	(-1.177)		
Industry FE	Yes	Yes	Yes	Yes		
Year FE	Yes	Yes	Yes	Yes		
R2	0.142	0.042	0.028	0.067		
Obs.	353	353 316		298		
Panel B: Implementation	n strategy using a lo	ong-time window.				
WSE	226.963 ***	-0.301 ***	-0.262 ***	-0.043		
	(3.825)	(-4.271)	(-2.902)	(-0.378)		
WSE x Post	-133.005 **	-0.159 *	-0.258	-0.141		
	(-2.044)	(-1.689)	(-1.296)	(-0.562)		
Industry FE	Yes	Yes	Yes	Yes		
Year FE	Yes	Yes	Yes	Yes		
R2	0.185	0.043	0.025	0.054		
Obs.	552	549	476	453		

Table C8. Effects of the 2014 regulatory change on the IPO underpricing, long-term underperformance and on the IPO size – alternative sample excluding 2008-2009 Financial Crisis.

The table reports results for the regression $Y_i = \beta_1 WSE_i + \beta_2 WSE_i \times Post_i + \gamma_t + \delta_j + \varepsilon_i$. The dependent variable is either: the IPO size (in million PLN), the first-day market-adjusted return, the 30-day market-adjusted return, or the 360-day market-adjusted return (all in percentages), and our unit of observation is a single IPO. *WSE* is an indicator variable that takes on a value of 1 for IPO listings in the WSE market, and 0 for the NC IPO listings. *Post* is an indicator variable that takes on a value of 1 for post-event IPO listings, and 0 for the pre-event IPO listings, *WSE* × *Post* is the variable of interest that captures the difference-in-differences effect. All regressions include industry and year fixed effects. Year fixed effects indicate the year of the IPO. *t*-statistics are reported in parentheses. Data are for the period February 4, 2011-February 3, 2017 for Panel A, and January 1, 2010-July 30, 2017 for Panel B. The sample excludes IPOs from the 2008 and 2009 corresponding to the Global Financial Crisis. The sample size varies depending on the length of the sample window as well as data requirements with respect to the dependent variable. For reference, we report the number of observations in the last row of each panel. Returns are in decimal format; IPO size is in million PLN. *, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively. We winsorize the observed IPO sizes at 0.5% and 99.5%.

	IPO Size	1st day market- adjusted return	30-day market- adjusted return	360-day market- adjusted return
Panel A: Implementatio	n strategy using a s	hort-time window.		
WSE	184.950 ***	-0.213 ***	-0.154	0.077
	(2.797)	(-4.367)	(-1.433)	(0.702)
WSE x Post	-127.814 *	-0.174 **	-0.872	-0.219
	(-1.895)	(-2.060)	(-1.620)	(-0.870)
Industry FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes 0.060
R2	0.136	0.040	0.032	
Obs.	454	453 400		378
Panel B: Implementation	n strategy using a l	ong-time window.		
WSE	208.709 ***	-0.232 ***	-0.235 ***	-0.122
	(3.598)	(-6.349)	(-2.723)	(-0.719)
WSE x Post	-120.096 *	-0.172 **	-0.782	-0.087
	(-1.904)	(-2.403)	(-1.551)	(-0.320)
Industry FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes
R2	0.142	0.039	0.030	0.028
Obs.	595	591	507	480

Table C9. Effects of the 2014 regulatory change on the IPO underpricing, long-term underperformance and on the IPO size – alternative threshold for the long-term return variables.

The table reports results for the regression $Y_i = \beta_1 WSE_i + \beta_2 WSE_i \times Post_i + \gamma_t + \delta_j + \varepsilon_i$. The dependent variable is either: the IPO size (in million PLN), the first-day market-adjusted return, the 30-day market-adjusted return, or the 360-day market-adjusted return (all in percentages), and our unit of observation is a single IPO. *WSE* is an indicator variable that takes on a value of 1 for IPO listings in the WSE market, and 0 for the NC IPO listings. *Post* is an indicator variable that takes on a value of 1 for post-event IPO listings, and 0 for the pre-event IPO listings, *WSE* × *Post* is the variable of interest that captures the difference-in-differences effect. All regressions include industry and year fixed effects. We adjust the threshold so that the post-intervention period includes all returns that could have been potentially affected, that is, observations of firms that listed during the pre-treatment period but which 30-day and 360-day interval for aftermarket performance measurement is also in the range of the post-treatment period. Year fixed effects indicate the year of the IPO. *t*-statistics are reported in parentheses. Data are for the period February 4, 2011-February 3, 2017 for Panel A, and January 1, 2008-July 30, 2017 for Panel B. The sample size varies depending on the length of the sample window as well as data requirements with respect to the dependent variable. For reference, we report the number of observations in the last row of each panel. Returns are in decimal format; IPO size is in million PLN. *, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively. We winsorize the observed IPO sizes at 0.5% and 99.5%.

	IPO Size	1st day market- adjusted return	30-day market- adjusted return	360-day market- adjusted return
Panel A: Implementatio	n strategy using a s	hort-time window.		
WSE	184.950 ***	-0.213 ***	-0.170	-0.174
	(2.797)	(-4.367)	(-1.517)	(-0.788)
WSE x Post	-127.814 *	-0.174 **	-0.809	0.180
	(-1.895)	(-2.060)	(-1.409)	(0.706)
Industry FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes
R2	0.136	0.040 0.033		0.028
Obs.	454	453 410		462
Panel B: Implementation	n strategy using a l	ong-time window.		
WSE	194.113 ***	-0.317 ***	-0.267 ***	-0.154
	(4.040)	(-5.382)	(-3.216)	(-0.905)
WSE x Post	-106.672 **	-0.103	-0.727	0.029
	(-1.968)	(-1.251)	(-1.396)	(0.125)
Industry FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes
R2	0.180	0.044	0.029	0.027
Obs.	706	702	603	576

Table C10. Effects of the 2014 regulatory change on the IPO underpricing, long-term underperformance and on the IPO size – alternative specification with interacted industry and Post effects.

The table reports results for the regression $Y_i = \beta_1 WSE_i + \beta_2 WSE_i \times Post_i + \gamma_t + \delta_j + \delta_j \times Post_i + \varepsilon_i$. The dependent variable is either: the IPO size (in million PLN), the first-day market-adjusted return, the 30-day market-adjusted return, or the 360-day market-adjusted return (all in percentages), and our unit of observation is a single IPO. *WSE* is an indicator variable that takes on a value of 1 for IPO listings in the WSE market, and 0 for the NC IPO listings. *Post* is an indicator variable that takes on a value of 1 for post-event IPO listings, and 0 for the pre-event IPO listings, *WSE* × *Post* is the variable of interest that captures the difference-in-differences effect. All regressions include industry and year fixed effects. Industry fixed effects are additionally interacted with the *Post* variable to control for the industry-specific impact of the regulation. Year fixed effects indicate the year of the IPO. *t*-statistics are reported in parentheses. Data are for the period February 4, 2011-February 3, 2017 for Panel A, and January 1, 2008-July 30, 2017 for Panel B. The sample size varies depending on the length of the sample window as well as data requirements with respect to the dependent variable. For reference, we report the number of observations in the last row of each panel. Returns are in decimal format; IPO size is in million PLN. *, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively. We winsorize the observed IPO sizes at 0.5% and 99.5%.

	IPO Size	1st day market- adjusted return	30-day market- adjusted return	360-day market- adjusted return		
Panel A: Implementation strategy using a short-time window.						
WSE	184.950 ***	-0.213 ***	-0.154	0.077		
	(2.797)	(-4.367)	(-1.433)	(0.702)		
WSE x Post	-127.814 *	-0.174 **	-0.872	-0.219		
	(-1.895)	(-2.060)	(-1.620)	(-0.870)		
Industry FE	Yes	Yes	Yes	Yes		
Industry FE x Post	Yes	Yes	Yes	Yes		
Year FE	Yes	Yes	Yes	Yes 0.060		
R2	0.136	0.040	0.032			
Obs.	454	453	400	378		
Panel B: Implementation	n strategy using a lo	ong-time window.				
WSE	194.113 ***	-0.317 ***	-0.271 ***	-0.135		
	(4.040)	(-5.382)	(-3.309)	(-0.930)		
WSE x Post	-106.672 **	-0.103	-0.737	-0.052		
	(-1.968)	(-1.251)	(-1.463)	(-0.202)		
Industry FE	Yes	Yes	Yes	Yes		
Industry FE x Post	Yes	Yes	Yes	Yes		
Year FE	Yes	Yes	Yes	Yes		
R2	0.180	0.044	0.029	0.027		
Obs.	706	702	603	576		

Table C11. Effects of the 2014 regulatory change on the IPO underpricing, long-term underperformance and on the IPO size – alternative regression specification with additional control variables and Post variable.

The table reports results for the regression $Y_i = \beta_1 WSE_i + \beta_2 Post_i + \beta_3 WSE_i \times Post_i + \beta_4 \log (Assets)_i + \beta_5 ROA_i + \beta_6 I(VC)_i + \gamma_t + \delta_j + \varepsilon_i$. The dependent variable is either: the IPO size (in million PLN), the first-day market-adjusted return, the 30-day market-adjusted return, or the 360-day market-adjusted return (all in percentages), and our unit of observation is a single IPO. *WSE* is an indicator variable that takes on a value of 1 for IPO listings in the WSE market, and 0 for the NC IPO listings. *Post* is an indicator variable that takes on a value of 1 for post-event IPO listings, and 0 for the pre-event IPO listings, $WSE \times Post$ is the variable of interest that captures the difference-in-differences effect. log (*Assets*) are the total assets in the last fiscal year prior to the IPO. *ROA* is the return on assets in the last fiscal year prior to the IPO, using the end-of-period value of total assets in the denominator. I(VC) is an indicator of whether an IPO was Venture Capital backed. All regressions include industry fixed effects. *t*-statistics are reported in parentheses. Data are for the period February 4, 2011-February 3, 2017 for Panel A, and January 1, 2008-July 30, 2017 for Panel B. The sample size varies depending on the length of the sample window as well as data requirements with respect to the dependent variable. For reference, we report the number of observations in the last row of each panel. Returns are in decimal format; IPO size is in million PLN. *, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively. We winsorize the observed IPO sizes at 0.5% and 99.5%.

	IPO Size	1st day market- adjusted return	30-day market- adjusted return	360-day market- adjusted return			
Panel A: Implementation strategy using a short-time window.							
WSE	35.816	-0.049	0.089	0.093			
	(1.530)	(-1.406)	(0.638)	(0.658)			
Post	-15.446	0.248 ***	1.013 *	0.291			
	(-0.953)	(3.590)	(1.892)	(1.074)			
WSE x Post	-98.095 *	-0.286 ***	-1.059 *	-0.221			
	(-1.861)	(-3.762)	(-1.789)	(-0.766)			
Controls	Yes	Yes	Yes	Yes			
Industry FE	Yes	Yes	Yes	Yes			
Year FE	No	No No					
R2	0.193	0.191 0.083		0.054			
Obs.	415	414	371	351			
Panel B: Implementation strategy using a long-time window.							
WSE	-35.017	-0.115 **	-0.035	0.064			
	(-1.126)	(-2.244)	(-0.402)	(0.575)			
Post	-7.696	0.176 **	0.892 *	0.270			
	(-0.423)	(2.573)	(1.743)	(1.027)			
WSE x Post	-68.249	-0.236 ***	-0.915 *	-0.201			
	(-1.460)	(-2.995)	(-1.687)	(-0.745)			
Controls	Yes	Yes	Yes	Yes			
Industry FE	Yes	Yes	Yes	Yes			
Year FE	No	No	No	No			
R2	0.292	0.091	0.070	0.039			
Obs.	643	639	556	531			

Table C12. Issuer effects of the 2014 regulatory change on the IPO underpricing, long-term underperformance and on the IPO size across different IPO size sample partitions – alternative specification with additional control variables.

The table reports results for the regression $Y_i = \beta_1 WSE_i + \beta_2 WSE_i \times Post_i + \beta_3 \log (Assets)_i + \beta_4 ROA_i + \beta_4 ROA_i$ $\beta_5 I(VC)_i + \gamma_t + \delta_i + \varepsilon_i$. We estimate the regression for the IPO size partitions. To construct the size partitions, we sort the sample based on IPO size each year. Prior to sorting, we exclude issuers with IPO size unavailable or reported as 0 by the exchange. Small issuers are issuers in the smallest size quartile (bottom 25% issuers). Big issuers are issuers in the largest size quartile (top 20% issuers). We also report intermediate portfolios (labelled portfolio 2 and 3) corresponding to the second and third quartile of IPO size. The dependent variable is either: the IPO size (in million PLN), the first-day market-adjusted return, the 30-day market-adjusted return, or the 360-day market-adjusted return (all in percentages), and our unit of observation is a single IPO. Log IPO size is the natural logarithm of IPO size. In the regression, WSE is an indicator variable that takes on a value of 1 for IPO listings in the WSE market, and 0 for the NC IPO listings. Post is an indicator variable that takes on a value of 1 for postevent IPO listings, and 0 for the pre-event IPO listings, $WSE \times Post$ is the variable of interest that captures the difference-in-differences effect. log (Assets) are the total assets in the last fiscal year prior to the IPO. ROA is the return on assets in the last fiscal year prior to the IPO, using the end-of-period value of total assets in the denominator. I(VC) is an indicator of whether an IPO was Venture Capital backed. All regressions include industry and year fixed effects. Year fixed effects indicate the year of the IPO. We only report difference-indifferences estimates, which are the coefficients on $WSE \times Post$. t-statistics are reported in parentheses. Data are for the period February 4, 2011-February 3, 2017 for Panel A, and January 1, 2008-July 30, 2017 for Panel B. The sample size varies depending on the length of the sample window as well as data requirements with respect to the dependent variable. Returns are in decimal format; IPO size is in million PLN. *, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively. We winsorize observed IPO sizes at 0.5% and 99.5%.

	IPO Size	Log IPO Size	1st day market- adjusted return	30-day market- adjusted return	360-day market- adjusted return
Panel A: Impleme	entation strategy	using a short-tim	e window.		
Small	-6.608	-0.538	-0.277 *	-0.233	-0.720
2	(-1.521)	(-1.516)	(-1.891)	(-0.630)	(-1.111)
	-4.520	0.028	-0.187	-0.151	0.430
3	(-0.713)	(0.204)	(-1.021)	(-0.395)	(1.526)
	-24.930 **	-0.610 **	-0.189	-0.198	-0.544
Big	(-2.310)	(-2.529)	(-1.150)	(-1.017)	(-0.885)
	-522.823 **	-0.587 *	-0.311 **	-0.685	0.152
Small	(-2.304)	(-1.823)	(-2.094)	(-1.474)	(0.460)
Panel B: Impleme	entation strategy	using a long-time	e window.		
Small	1.395	0.234	-0.223	-0.171	-0.705
	(0.402)	(0.579)	(-1.414)	(-0.481)	(-1.154)
2	2.502	0.277 *	-0.200	0.135	0.411
	(0.491)	(1.865)	(-0.787)	(0.389)	(1.597)
3	29.954	-0.103	-0.049	-0.246 *	-0.343
	(1.122)	(-0.398)	(-0.328)	(-1.715)	(-0.669)
Big	-334.022 * (-1.663)	-0.397 (-1.303)	-0.252 ** (-2.068)	-0.617 (-1.444)	0.186 (0.596)